GENERAL	PUBLIC	TARIFF	OF APM	TERMINALS	CALLAO	S.A

v 14.7 Special tariff (USD) Item Services description Nature Unit of charge Tariff (USD)

SECTION 1 Standard Service Unit of charge Vessel (Tariff) Vessel (IGV) Cargo (Tariff) Cargo (IGV) Vessel (Tariff) Vessel (IGV) Cargo (Tariff) Cargo (IGV) Section 1.1 Use or netal of beth (Tariff per hour or fraction hour) (n1)
Loading and Unloading Containers - STS Gantry Cranes (n2 and n3)
Unloading Ful 20 Foot Container with dry cargo
Unloading Ful 20 Foot Container with dry cargo
Loading Ful 20 Foot Container with dry cargo 2.00 0% Rate 0% Rate Regulated Total LoA (m) x Hours 1.50 1.1.2 1.1.2.1 Per move 12.20 0% Rate 217.50 39.15 11.45 0% Rate 206.25 37.13 1.1.2.2 Per move Per move 0% Rate 0% Rate 401.00 217.50 72.18 39.15 0% Rate 345.95 156.20 62.27 28.12 Loading Full 40 Foot Container with dry cargo 72.18 1.1.2.4 Per move 12.20 0% Rate 401.00 11.45 277.10 49.88 Unloading Full 40 Foot Container with refrigerated cargo
Unloading Full 40 Foot Container with refrigerated cargo
Loading Full 40 Foot Container with refrigerated cargo
Loading Full 40 Foot Container with refrigerated cargo Per move Per move 1125 12 20 217 50 39 15 11 45 217 40 39 13 1.1.2.6 1.1.2.7 401.00 217.50 72.18 11.45 11.45 401.00 217.40 Per move 12.20 39.15 39.13 Per move Per move Per move 1.1.2.8 12.20 401.00 72.18 11.45 401.00 72.18 Loading Full Marging Empty 20 Foot Container with refrigerated cargo
Loading and Uniosania (Empty 20 Foot Container
Loading and Uniosania (Empty 40 Foot Container
Loading and Uniosania (Empty 40 Foot Container
Complete cycle 20 Foot Transhipment Container (full or empty) within the Terminal (n4)
Complete cycle 40 Foot Transhipment Container (full or empty) within the Terminal (n4)
Loading and Uniosania (Empty Container Withhout STS Gantry Cranes (n3 and n5)
Loading and Uniosania (Full 20 Foot Container with dry cargo
Loading and Uniosania Full 20 Foot Container with dry cargo
Loading and Uniosania Full 20 Foot Container with dry cargo
Loading and Uniosania Full 20 Foot Container with dry cargo 1.1.2.9 202.00 0% Rate 147 30 0% Rate 1.1.2.10 1.1.2.11 0% Rate 0% Rate 217.50 127.50 Per complete cycle 229.00 0% Rate 1.1.2.12 Per complete cycle 349.00 0% Rate 190.60 0% Rate 1.1.3 1.1.3.1 1.1.3.2 Per move 6.30 0% Rate 0% Rate 35.21 6.00 0% Rate 0% Rate 141.20 25.42 58.72 6.00 Per move 1.1.3.3 Loading and Unloading Full 20 Foot Container with refrigerated cargo Per move 6.30 0% Rate 195.60 35.21 6.00 0% Rate Loading and Unloading Full 40 Foot Container with refrigerated cargo Loading and Unloading Full 40 Foot Container with refrigerated cargo Loading and Unloading Empty 20 Foot Container Loading and Unloading Empty 40 Foot Container Per move Per move Per move 1134 6.30 0% Rate 326 20 58 72 6.00 0% Rate Regulated 1.1.3.5 1.1.3.6 162.40 253.40 0% Rate 0% Rate 0% Rate 0% Rate 180.30 1.1.3.2 Coasing and unasang Empty au root Container (full or empty) within the Terminal (n4)
1.1.3.8 Complete cycle 20 Foot Transhipment Container (full or empty) within the Terminal (n4)
1.1.3.8 Complete cycle 20 Foot Transhipment Container (full or empty) within the Terminal (n4)
(n1) including doding and undoding. The total length of the ship to be confirmed by the "Ship's Particules". The use of the beth as from the receipt of the first line of the ve
(n2). The containers whose dimensions are less than 20-foot containers will be charged with 20-foot container tantf. The containers whose dimensions are less than 20-foot Per complete cycle Per complete cycle 182.00 0% Rate 112.80 0% Rate 271.90 0% Rate 168 70 0% Rate

(n5): This is applicable only in those berths which do not have STS gantry cranes and in those cases when loading/unloading operations of containers will be carried out without STS gantry cranes.

	Section 1.2	Special Services to the Vessel (n6)	Nature	Unit of charge	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff) Cargo	(IGV) Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)
Marche   M	1.2.1										
Maries of Continue (mpt)		Movement of ISO hatches	Regulated	Per move	0.00	0% Rate					
Decision was not 20 fort attended concinence   Proposed   Propos	1.2.2	Restow of containers (n6)									
Register with 40 for clamped container   Register   Procuration   Programme   5,00   On Side				Per container	0.00	0% Rate					
New York or 20 for foreigned contemps   New York or											
Section of comparing replaceshing with Misching Misching Comes (MIC)   No. Regulated   Pro-Count With Misching Misching Misching Comes (MIC)   No. Regulated   Pro-Count With Misching Misching Misching Comes (MIC)   No. Regulated   Pro-Count With Misching Misching Comes (MIC)   No			Regulated								
1.32   Super yelf-to increase protective productions with which feather consequence protection price part and in detail feather											
1.2.1   1.2.2   1.2.				rei containei	0.00	U% Rate					
1.2.6   Live of Cognition Association (Associations)   Proposition (Association)   Proposition)   Proposition (Association)   Proposition (A			Non Donated	D O (Oh)6	0.500.00	00/ D-1-					
1.2.1.1   1.2.1.2   1.2.			Non Regulated	Per Crane/Snitt	2,500.00	U% Rate					
Per TEURing   10.0   0.0   Rea			Developed		Eee						
1.2.14   Day   Price per day    Per TEUrlay   Per TEURla			Regulateu	Dor TELlidou							
Per TSUARY   Price part day			Non Regulated								
New Foundation   New			Non regulated								
1.2.15   1.2.25   1				1 61 TLO/day	10.10	07011410					
1.2.2.5   Day 7-1 Per 10 July 1			Regulated		Fre	ie.					
12.53   Day 7 (Price per day)			- Togulateu	Per TFU/day							
12.54   Use of Companied Price per day)   Fee Tible   Fee Tible			Non Regulated								
March Operational Name - Transformer Full Conditions:   1.26.1											
About Fire the michael on Student's Service) (n1)											
Day 3 - 4 (Pinca per day)			Regulated		Fre	10					
Day 5 - 6   Pirco per day				Per TEU/day	17.10	0% Rate					
1.27   48 hose price microaction of Secretary Containers   1.27   49 hose per day    17.10   0% Rate   12.27   12.29			Non Regulated			0% Rate					
A Price of Price per day   Per TEUday   Pe	1.2.6.4	Day 7 onwards (Price per day)		Per TEU/day	33.00	0% Rate					
Day 3 - 4 Ploto per day	1.2.7	Use of Operational Area - Transhipment Empty Containers									
Day 5 - 6 (Price per day)   Day 5 - 6 (Price per day)	1.2.7.1	48 hours (Free time included in Standard Service) (n11)	Regulated		Fre	ië					
Day 7 owners   Price per day	1.2.7.2	Day: 3 – 4 (Price per day)		Per TEU/day	17.10	0% Rate					
Use of Operational Nate for Empty good - Empty containers toward you off (2)   Non Regulated   Per EU   Per agreement	1.2.7.3	Day: 5 – 6 (Price per day)	Non Regulated	Per TEU/day	26.80	0% Rate					
1.2.6   Sept Container and For arroy containers from empty pool (nt/2)   Per agreement	1.2.7.4	Day 7 onwards (Price per day)		Per TEU/day	33.00	0% Rate					
Loading Unitodating and delivery of non ISO/OGG (Additional) Container 20 and 40 foot (n7)   Per TEU   260.00   0% Rate			Non Regulated	Per TEU	Per agre	ement					
Testiment of Nazionous cargo containers MNO class 1 (m/3)   Per TEU   260,00   0% Rate											
Teatment of Hazadous cargo containers (MO classes (F1))   Per TEU   120,00   On Fate			Regulated								
Per EU   12.00   0% Rate											
1.2.15   Treatment of Hazardous cargo cher MD Classes (n13)											
1.2.9   Supply of special equipment for handing 20 containers with oversized cargo in the container yard   Containers with oversized cargo in the container yard     1.2.9   Supply of special equipment for handing 40 containers with oversized cargo in the container yard     1.2.9   Supply of special equipment for handing 40 containers with oversized cargo in the container yard     1.2.9   Supply of special equipment for handing 40 containers with oversized cargo in the container yard     1.2.10   Complementary Services to street/oring/instruction of disconnection and disconnection service for refer containers on board vessels (n15)     1.2.11   Instruction and disconnection service for refer containers on board vessels (n15)     1.2.11   Energy - container 20 and 40 bod (n0 y n16)     1.2.11   Energy - container 20 and 40 bod (n0 y n16)     1.2.11   Energy - container 20 and 40 bod (n0 y n16)     1.2.11   Energy - container 20 and 40 bod (n0 y n16)     1.2.11   Energy - container 20 and 40 bod (n0 y n16)     1.2.11   Energy - container 20 and 40 bod (n0 y n16)     1.2.11   Energy - container 20 and 40 bod (n0 y n16)     1.2.11   Energy - container 20 and 40 bod (n0 y n16)     1.2.11   Energy - container 20 and 40 bod (n0 y n16)     1.2.11   Energy - container 20 and 40 bod (n0 y n16)     1.2.11   Energy - container 20 and 40 bod (n0 y n16)     1.2.11   Energy - container 20 and 40 bod (n0 y n16)     1.2.11   Energy - container 20 and 40 bod (n0 y n16)     1.2.11   Energy - container 20 and 40 bod (n0 y n16)     1.2.11   Energy - container 20 and 40 bod (n0 y n16)     1.2.11   Energy - container 20 and 40 bod (n0 y n16)     1.2.11   Energy - container 20 and 40 bod (n0 y n16)     1.2.11   Energy - container 20 and 40 bod (n0 y n16)     1.2.11   Energy - container 20 and 40 bod (n0 y n16)     1.2.12   Energy - container 20 and 40 bod (n0 y n16)     1.2.13   Energy - container 20 and 40 bod (n0 y n16)     1.2.14   Energy - container 20 and 40 bod (n0 y n16)     1.2.15   Energy - container 20 and 40 bod (n0 y n16)											
1.2.9.2   Supply of special equipment for handling 47 containers with oversized cargo: use of operational rate (n14)			Non Regulated								
1.2.9.   High Carcinaries Set (1.2.1)   Per container (2.2.1)   Per containe											
12.19   High Cube Containers 95' (n14)   Per container 20.00   O% Rate											
1.2.10   Complementary Services to street/doring/instreet/doring											
1.2.10.1   Connection and disconnection service for reefer containers on-board vessels (n15)   Non Regulated   Per container-operation   9.00   0% Rate				T OF CONTAINO	20.00	07011010					
			Non Regulated	Per container-operation	9.00	0% Rate					
1.2.11.1   Energy - container 20 and 40 bott (ny n 16)   Regulated				22.12.12.22.27.0001	2.00						
12-112   Inspection and monitoring -container 20 and 40 foot (nt?)			David : :	Per Container/hour	0.00	0% Rate					
12.11.3   Energy - container 20 and 40 bodt (nt7)   Non Regulated   Per Container 20 and 40 bodt (nt7)   Non Regulated   Per event   27.20   0% Rate			Regulated								
1.2.11.2   Inspection and monitoring - container 20 and 40 foot (n17)   Per event   27.0   0% Rate			Non Doordoted								
			Non Regulated								
12.13   Sasification of a full container reefer - all traffics (n19)   Per container   460.0   82.80     12.13   Supply of gases N2   Regulated   Per container   411.00   73.98     12.13   Supply of gases N2   Regulated   Per container   192.00   34.56     12.14   Regulated   Per container   192.00   34.56     12.14   Download and sending information of reefer container parameters   Regulated   Per container   30.00   5.40     12.15   Additional inspection or revision of seals     12.15   Additional inspection or revision of seals (n20)   Non Regulated   Per seal   60.00   0% Rate     12.16   Taction Service for Inter-terminal Transphipment Containers (n21)   Non Regulated   Per container   40.00   0% Rate	1.2.12										
1.2   1.3   Supply of gases N2 y COZ   Regulated   Per container   480,00   82,80     1.2   1.3   Supply of gases N2   Regulated   Per container   411,00   73,38     1.2   1.3   Supply of gases COZ   Per container   192,00   34,56     1.2   1.4   Refer container data download   Per container   192,00   34,56     1.2   1.4   Supply of gases ROZ   Regulated   Per container   192,00   34,56     1.2   1.4   Supply of gases ROZ   Regulated   Per container   192,00   34,56     1.2   1.4   Supply of gases ROZ   Regulated   Per container   192,00   34,56     1.2   1.4   Supply of gases ROZ   Regulated   Per container   30,00   5,40     1.2   1.5   Reference   Regulated   Per seal   60,00   0% Rate     1.2   1.4   Supply of gases ROZ   Regulated   Per seal   60,00   0% Rate     1.2   1.4   Supply of gases ROZ   Regulated   Per container   40,00   0% Rate     1.2   1.4   Supply of gases ROZ   Regulated   Per container   40,00   0% Rate     1.2   1.4   Supply of gases ROZ   Regulated   Per container   40,00   0% Rate     1.2   1.3   Reference   Regulated   Per container   40,00   0% Rate     1.2   1.3   Reference   Regulated   Per container   40,00   0% Rate     1.2   1.3   Reference   Regulated   Per container   40,00   0% Rate     1.2   1.3   Reference   Regulated   Per container   40,00   0% Rate     1.2   1.3   Reference   Regulated   Per container   40,00   0% Rate     1.2   1.3   Reference   Regulated   Per container   40,00   0% Rate     1.2   1.3   Reference   Regulated   Per container   40,00   0% Rate     1.2   1.3   Reference   Regulated   Regulated   Per container   40,00   0% Rate     1.2   1.3   Reference   Regulated   Regulated	1.2.12.1		Non Regulated	Per container	77.10	0% Rate					
1.2.13.2   Supply of gases N2	1.2.13	Gasification of a full container reefer - all traffics (n19)									
1.2.13.   Supply of gases CO2											
1.2.14.   Download and sending information of reefer container parameters   Regulated   Per container   30.00   5.40     1.2.15.   Additional inspection or revision of seals     1.2.15.   Additional inspection or revision of seals   (20)   Non Regulated   Per seal   60.00   0% Rate     1.2.16.   Traction Service for Inter-terminal Transhipment Containers   Non Regulated   Per container   40.00   0% Rate     1.2.16.   Inter-terminal Transhipment Containers (21)   Non Regulated   Per container   40.00   0% Rate			Regulated								
1.2.14.1 Download and sending information of reefer container parameters 9.00 5.40  1.2.15 Additional inspection or revision of reefer sending information of reefer container parameters 9.00 0 % Rate 9.10 1.2.15 Additional inspection or revision of revision of revision of seals (r20) 9.00 0 % Rate 9.10 1.2.15 Non Regulated 9.00 0 0 % Rate 9.10 1.2.15 Non Regulated 9.00 0 0 % Rate 9.10 1.2.15 Non Regulated 9.00 0 0 % Rate 9.10 1.2.15 Non Regulated 9.00 0 0 % Rate 9.10 1.2.15 Non Regulated 9.00 0 0 % Rate 9.10 1.2.15 Non Regulated 9.00 0 0 0 % Rate 9.10 1.2.15 Non Regulated 9.00 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0				Per container	192.00	34.56					
1.2.15.         Additional inspection of seals         Non Regulated         Per seal         60.00         0% Rate           1.2.15.1         Traction Service for Inter-terminal Transhipment Containers         Non Regulated         Per container         40.00         0% Rate           1.2.16.1         Inter-terminal Transhipment Containers (n2!)         Non Regulated         Per container         40.00         0% Rate											
1.2.15.1         Additional inspection or revision of seals (r/C0)         Non Regulated         Per seal         60.00         0% Rate           1.2.16.1         Traction Service for Inter-terminal Transhipment Containers         40.00         0% Rate           1.2.16.1         Inter-terminal Transport (Transhipment Containers (r/C1)         Non Regulated         Per container         40.00         0% Rate			Regulated	Per container	30.00	5.40					
1.2.16.1     Traction Service for Inter-terminal Transhipment Containers       1.2.16.1 Inter-terminal Transport of Transhipment Containers (n21)     Non Regulated       Per container     40.00       O% Rate											
1.2.16.1 Inter-terminal Transport of Transhipment Containers (n²t)  Non Regulated  Per container  40.00 0% Rate			Non Regulated	Per seal	60.00	0% Rate					
			Non Regulated	Per container	40.00	0% Rate					

(n7): Applicable to all users that necessarily require the service(s), according to conditions established in the Tariffs Regulation and Commercial Policy of APM Terminals Callao S.A.

(rid): Charge of this price is applicable whenever service is required by users and is aimed to increase productivity of partial reminals Calloo S.A. and with no user request, productivity increase cost is under resp. PMT reminals Calloo S.A. and with no user request, productivity increase cost is under resp. PMT reminals Calloo.

(n9): Free time to run as from the end of complete discharge operation of the ship

[n10]: 72 Free hours to run as from the period cargo is located in the Terminal yard for shipment. Storage time ends with vessel ETB sent to the users. For purposes of computation of storage free time storage and billable storage, it should be taken in consideration is the so-called Billable ETB' that is published in the 'Container Ship Programme' option on our website and defined in the Tarriffs Regulation and Commercial Policy of APM Terminals Callao S.A.

(n11): Free time to run as from the end of complete discharge operation of the vessel or from the moment the container is located in the Terminal yard for shipment. Free time does not apply to full transshipment containers with oversized cargo; that is, for this type of container, the use of operating area is billed from the first day

(11) in the event that a container carries several dasses of IMO cago, the higher price amount of IMO classes will be applicable as one sole charge. Price of subsection 12.9.5 does not apply for ammonium nitrate IMO Class 51, unless the user requires the service. Price of subsection 12.9.4 does not apply for containers of Fishmeal IMO class 5. (n14). This service covers additional capacity of containers yard or stacks of empty containers (all Itaffic) and full transhipment containers, although in the case of containers with oversized days at the Terminal.

(n15): This price is applicable to all discharged or shipped containers through the Terminal including restow containers (via quay and in the same hold), that require such service. This is applicable for this operation, whether it is connection or disco (n16): Teriff applicable until day 10th of storage (inclusive).

(n17): Price applicable as from 11th day onward.

(n18): This is applicable to re-stow containers via quay, including inspection and monitoring if necessary. This is an one time payment to be applicable for total energy supply period provided to re-stow reefer containers

(1/10). This is applicable to throse cases whenever there is a difference between seal information provided by APM Terminals Callao S.A. and information from the user, for additional services requested by the user. In the event APM Terminals Callao S.A. originates seal discrepancy no charge will be applicable. Not applicable as well for loading controls. (n21): Service that involves moving the container between port terminals from the North Terminal to the South Terminal whenever a container is unloaded via the North Terminal and it is required to be shipped via the South Terminal and/or viceversa.

Special Service of Use of Operational Area (Port Terminal) - Services to the Cargo
Use of Operational Area - Import Full Containers (accept Transhipment) (n22)
48 hours to import and/or discharge container (Free time included in Standard Service) (n23) Section 1.3 Nature Unit of charge Vessel (Tariff) Vessel (IGV) Cargo (Tariff) Cargo (IGV) Vessel (Tariff) Vessel (IGV) Cargo (Tariff) Cargo (IGV) 1.3.1 48 hours to import and/or discharge Day: 3 – 6 (Tariff per day) 1311 Day 7 (Price per day) container 20 and 40 foot 1.3.1.3 Per TEU/day 90.00 16.20 1.3.1.4 Days: 8 - 10 (Price per day) Per TEU/day 90.00 16.20 Non Regulated 1.3.1.5 1.3.1.6 Days: 11 – 20 (Price per day) Days: 21 – 28 (Price per day) Per TEU/day 90.00 16.20 1.3.1.7 Day: 29 onwards (Price per day) Per TEU/day 90.00 16.20 1,3,2 Use of Operational Area - Export Full Containers (except Transhipment) (n22) 72 hours to export and/or loading container (Free time) (n Days: 4 – 6 (Tariff per day) 1.3.2.3 Day 7 (Price per day) container 20 and 40 foot Por TEU 90.10 16.22

1.3.2.4	Days: 8 – 10 (Price per day)	Non Regulated	Per TEU/day	44.50	8.01	
1.3.2.5	Days: 11 – 20 (Price per day)	Non Regulateu	Per TEU/day	44.50	8.01	
1.3.2.6	Days: 21 – 28 (Price per day)		Per TEU/day	44.50	8.01	
1.3.2.7	Day: 29 onwards (Price per day)		Per TEU/day	44.50	8.01	

: Tariffs and prices set forth in sub-section 1.3.1 and 1.3.2 are applicable to those containers that have been discharged/loaded via the North Terminal as per conditions esta

(n22); Free time to run as from the end of complete cargo discharge operation of the vessel. Calculation of use of operational area starts from the date and hour of complete cargo discharge from vessel and finishes when container is withdrawn from the yeard. Free time does not apoply to full transnippment containers with oversized cargo, that is, for this type of container, the use of operating area is billed from the first day see's Estimated Time of Berthing (ETB) to be communicated to users. For purposes of computation of storage free time storage and billable storage, it should be to with oversized cargo; that is, for this type of container, the use of operating area is billed from the first day.

Section 1.4	Temporary Depot Special Service - Services to the Cargo	Nature	Unit of charge	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)
1.4.1	Temporary Depot Integrated Service for dry import containers (n25)										
1.4.1.1	20' Container - Includes use of area until the 10th day		Per container			334.75	60.26				
1.4.1.2	40' Container - Includes use of area until the 10th day		Per container			489.95	88.19				
1.4.1.3	Use of area during days 11 – 20 (Price per day) (n26)	Non Regulated	Per TEU/day			20.00	3.60				
1.4.1.4	Use of area during days 21 – 28 (Price per day) (n26)		Per TEU/day			25.00	4.50				
1.4.1.5	Use of area from 29th day onwards and other services (n26)			See section 1.3.1.	7 and others from	section 1.5					
1.4.2	Temporary Depot Integrated Service for dry export containers (n27)										
1.4.2.1	20' Container - Includes use of area until the 7th day		Per container			243.10	43.76				
1.4.2.2	40' Container - Includes use of area until the 7th day		Per container			375.70	67.63				
1.4.2.3	Use of area days 8 – 10 (Price per day) (n28)	Non Regulated	Per TEU/day			10.00	1.80				
1.4.2.4	Use of area days 11 – 20 (Price per day) (n28)	14011 (Yegulateu	Per TEU/day			14.00	2.52				
1.4.2.5	Use of area during days 21 – 28 (Price per day) (n28)		Per TEU/day			20.00	3.60				
1.4.2.6	Use of area from 29th day onwards and other services (n28)			See section 1.3.2.	7 and others from	section 1.5					
1.4.3	Temporary Depot integrated service for reefer import containers (n29)										
1.4.3.1	20' Container - Includes use of area until the 7th		Per container			397.90	71.62				
1.4.3.2	40' Container - Includes use of area until the 7th	Non Regulated	Per container			570.40	102.67				
1.4.3.3	Use of area during days 8 – 10 (Price per day) (n30)	rvoii rvegulateu	Per TEU/day			15.00	2.70				
1.4.3.4	Use of area from day 11 onwards and other services (n30)		See se	ction 1.4.1.3, 1.4.1.4	and 1.3.1.7 and ot	hers from section 1.5	5				
1.4.4	Temporary Depot integrated service for reefer export containers (n31)										
1.4.4.1	20' Container - Includes use of area until the 7th		Per Container			364.90	65.68				
1.4.4.2	40' Container - Includes use of area until the 7th	Non Regulated	Per Container			518.40	93.31				
1.4.4.3	Use of area from day 8 onwards and other services (n32)		See	section 1.4.2.3 to 1.4.	2.5, 1.3.2.7 and of	hers of section 1.5					

(n/25). This shall be applicable whenever APM Terminatic Gallation S.A. has been nominated as Temporary Open (2014). Service includes Cargo Portion of Standard Service, use of operational area until the 10th day, documentation revision, temporary depot documentation or evision, temporary depot documentation evision, temporary depot docum

(n/20); This shall be applicable whenever APM Terminals Callao S.A. has been nominated as Temporary depot documentation emission (volante), customs transmission. Use of operational area until the 7th day, energy supply to reefer containers until the 7th day, inspection and monitoring until the 7th day, documentation revision, temporary depot documentation emission (volante), customs transmission. Use of operational area is calculated in from the date of complete cargo discharge of the vessel until the container is withdrawn from the container yard. First two days (48 hours) are free of charge and are included within the standard service.

(n30): Price of section 1.4.3.3 is applicable to use of operational area from days 8 to 10 of Temporary Deport, price of section 1.5.2.4 is applicable for days 11 to 20; price of section 1.4.1.4 is applicable for days 21 to 28; and price of section 1.3.1.7 is applicable for days 29 onwards. Price of section 1.5.2.3 is applicable for energy supply for day 8th onwards and price of section 1.5.2.4 is applicable for days 91 to 20; price of section 1.5.1.1.5.3, 1.5.4 and 1.5.5. are applicable independently from the number of storage days.

(n3)) This shall be applicable whenever APM Terminals Callar S.A. has been nominated as Temporary Seport GOMs, Service includes Cargo Portion of Standard Service, use of operational area until the 'Ph day, energy supply to reefer containers until the 'Ph day, inspection and monitoring until the 'Ph day, documentation revision, temporary depot documentation emission (volante), customs transmission. Use of operational area is calculated from the date the container is placed in the container are just until the Estimated Time of Senting (ETB) of the vessel. For purposes of computation of storage free time storage and billable storage, it should be taken in consideration is the so-called 'Billable ETB' that is published in the 'Container' Ship Porgramme' option on our webste and defined in the Tariffs Regulation and Commercial Policy of APM Terminals Callac S.A. First two days (48 hours) are free of change and are included within the standard service.

(nS2): Price of section 1.4.2.3 is applicable to use of operational area from days 8 to 10 of Temporary Depot price of section 1.5.2.4 is applicable for days 21 to 28; and price of section 1.3.2.7 is applicable for days 29 onwards. Price of section 1.5.2.3, is applicable for energy supply for day 8th onwards and price of section 1.5.2.4 is applicable for days 21 to 28; and price of section 1.3.2.7 is applicable for days 29 onwards. Price of section 1.5.3, is applicable for energy supply for day 8th onwards and price of section 1.5.3.4 and 1.5.3. are applicable independently from the number of storage days.

ction 1.5	Other Special Services for Containers (Port Terminal & Tempory Depot) - Services to the Cargo	Nature	Unit of charge	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)	Vessel (Tariff) Vessel (IGV) Cargo (Tariff)
1.5.1 Spe	ecial Cargo							
.5.1.1 Tre	eatment of Hazardous cargo containers IMO class 1 (n33		Per TEU			270.40	48.67	
1.5.1.2 Tre	eatment of Hazardous cargo containers IMO class 5.2, class 6.2 and class 7 (n33)		Per TEU			270.40	48.67	
1.5.1.3 Tre	eatment of Hazardous cargo IMO 9 (n33 and n34)		Per TEU			125.00	22.50	
1.5.1.4 Tre	eatment of Hazardous cargo other IMO classes (n33)	Non Regulated	Per TEU			159.00	28.62	
1.5.1.5 Sup	pply of special equipment for handling 20' containers with oversized cargo in the container yard	Non Regulated	Container-Move			169.00	30.42	
1.5.1.6 Sup	pply of special equipment for handling 40' containers with oversized cargo in the container yard		Container-Move			330.00	59.40	
1.5.1.7 Cor	intainers with oversized cargo - use of operational area (n35)		Per container/day			25.00	4.50	
1.5.1.8 Hig	gh Cube Containers 9'6" (n35)		Container			23.00	4.14	
1.5.2 Rec	efer Container							
1.5.2.1 Ene	ergy - 20' and 40' container (n36 and n37)	Regulated	Per container/hour			0.00	0.00	
1.5.2.2 Insp	spection and monitoring - 20' and 40' container (n36 and n37)	Regulateu	Per container/day			0.00	0.00	
1.5.2.3 Ene	ergy - 20' and 40' container (Price per day or fraction day) (n38)		Per container/day			78.00	14.04	
1.5.2.4 Insp	spection and monitoring - 20' and 40' container (n38)	Non Regulated	Per event			28.30	5.09	
1.5.2.5 Pre	e-cooling empty container (n39)	NOII Negulatéd	Per Container			46.00	8.28	
1.5.2.6 Pro	ovision of cold antechamber (n40)		Per Unit/day			250.00	45.00	
1.5.2.7 Pre	e-cooling empty container (n39)	Regulated	Per Container			46.00	8.28	
1.5.2.8 Ass	sembly and disassembly of clip system on reefer Gensets (n36)	Regulateu	Per complete cycle			0.00	0.00	
1.5.3 Car	rgo Handling and horizontal movement operations							
1.5.3.1 Add	ditional Movement to Standard Service in the terminal as per request of the user or authorities (n36)		Per Container			0.00	0.00	
1.5.3.2 Car	rgo inspection with groups o gangs - 20' and 40' container (n36)	Regulated	Per Container			0.00	0.00	
.5.3.3 Car	rgo inspection with forklifts - 20' and 40' container (n36)		Per Container			0.00	0.00	
1.5.3.4 Cor	visolidation/deconsolidation of containers with dry cargo (with forklift) (n36)		Per Container			176.00	31.68	
1.5.3.5 Cor	insolidation/deconsolidation of containers with dry cargo (with crew or crew-forklift) (n36)		Per Container			200.00	36.00	
1.5.3.6 Cor	resolidation/deconsolidation of containers with dry cargo (with coil handling forklift or another type of special forklift) (n36)		Per Container			230.00	41.40	
1.5.3.7 Cor	resolidation/deconsolidation of containers with dry cargo (with forklift) including Gate In/Gate Out of empty container (n36)		Per Container			230.00	41.40	
1.5.3.8 Cor	insolidation/deconsolidation of containers with dry cargo (with crew or crew-forklift) including Gate In/Gate Out of empty container (n36)		Per Container			250.00	45.00	
1.5.3.9 Cor	resolidation/deconsolidation of containers with dry cargo (with coil handling forklift or another type of special forklift) including Gate In/Gate Out of empty container (n36)	Non Regulated	Per Container			280.00	50.40	
1.5.3.10 Cor	visolidation/deconsolidation of containers with refrigerated cargo		Per Container			780.00	140.40	
1.5.3.11 Cor	resolidation/deconsolidation of containers with refrigerated cargo including Gate In/Gate Out of empty containers		Per Container			980.00	176.40	
1.5.3.12 Cor	resolidation/deconsolidation of containers with Mobile Harbour Cranes (Price per shift or fraction shift) (n41)		Per shift			1,250.00	225.00	
1.5.3.13 Par	rtial consolidation/deconsolidation of containers with dry cargo		Per Container			100.00	18.00	
1.5.3.14 Dep	palletizing/palletizing (removal of straps and/or strapping) (n42)		Per Pallet			10.00	1.80	
1.5.4 Oth	her Services							
1.5.4.1 Bre	eakdown as per instructions of freight forward agent (n43)		Per B/L			25.00	4.50	
1.5.4.2 Bill	of lading breakdown (n43)		Per breakdown			25.00	4.50	
1.5.4.3 Sw	veeping of empty container		Per container			7.00	1.26	
1.5.4.4 Sim	mple cleaning of container		Per container			26.00	4.68	
1.5.4.5 Che	remical cleaning of container		Per container			52.00	9.36	
1.5.4.6 Cor	ntainer repair	Non Regulated	Per Activity			Perd		
1.5.4.7 Sea	al placement services		Per seal			13.00	2.34	
1.5.4.8 Add	ditional inspection or revision of seals (n20)		Per seal			60.00	10.80	
1.5.4.9 Lab	bels Placing or Removal Services (n44)		Per container			10.00	1.80	
1.5.4.10 Add	ditional weighing of containers (n45)		Per container			90.00	16.20	
1.5.4.11 Sea	al breaking and/or container opening without cargo handling (n40)		Per container			40.00	7.20	
1.5.5 Dep	pot for Empty Containers							
1.5.5.1 Gat	te In (n46)	Non Regulated	Per container			240.00	43.20	
1.5.5.2 Gat	ite Out (n46)	NUII Negulated	Per container			272.00	48.96	

(n34). The price of this service (subsection 1.5.1.3) does not apply for Fishmeal - IMO Class 9 previously declared as such in the Container Announcement List (CAL) for shipping as per procedures.

(RSS): This service covers additional capacity of container yard or stacks of full containers (except transformer). In the case of high-cube containers the price is independent from the number of storage days at the Terminal and this is not applicable for containers of Temporary Depot service (section 1.4 services). In the case of containers with oversized cargo, the use of seria is billed from the first day (does not include free (n36): Applicable to all users that necessarily require the service(s), according to conditions established in the Tariffs Regulation and Commercial Policy of APM Terminals Callao S.A.

(GZT): This is applicable until 6th day of storage also. This tarff is applicable to all containers that are loaded or discharged via the Multipurpose North Terminal, except for transhoment and re-stow containers.

(rS8): This tariff is applicable from day 7 onwards for Port Terminal services (when APMITC has not been appointed Temporary Depot). Whenever APMITC is appointed as Temporary Depot, price is applicable from day 8th onwards.

(xid): This service refer to connecting of focus reports and connecting of the empty refer container and supplying energy to sed container, in order to adjust to the cold temperature required by the container and supplying energy to sed container. This service is a price (not regulated) when the pre-cooled empty container is assigned to serve all those goods that come from (or will move bounded) a full container unbased or loaded of vicaded by the Port Terminal (section 1.5.2.5). In those cases that the meritainties was unbased or to added at break but, cargo (frecen fish and others) the regulated service tariff applies (section 1.5.2.7).

[(xid): This applicable by all users (considered or loaded at break but, cargo (frecen fish and others) the regulated service tariff applies (section 1.5.2.7).

(A1): Service includes the use of Mobile Harbour Cranes por a period of 4 hours. For deconsolidations, service also includes the separation of the congree (or representative) in the event of consolidations, it completes the recipility of the congree (or representative) and the consolidation of the cargo or a placement inside the container (filst rack or another). This is subject to Mobile Harbour Cranes availability.

((A2): Price application is a large to a large to a large to a placement inside the outsider (filst rack or another). This is subject to Mobile Harbour Cranes availability.

((A2): Price application is a large to a large size of the company or a placement inside the outsider (filst rack or another). This is subject to Mobile Harbour Cranes availability.

((A2): Price application is a large size of the surpose of the price of the service or case where the loss better that cargos is a large size of the surpose of size of size

cases (either prior, by red channel or requested by BOE personnel).
(n43): It is charged to each breakdown B/L mother, B/L son or B/L grandson

(n/4): Respective labels shall be placed according to cargo dassification as per IMDG code for export operations. For import operations, placing of label shall take place whenever a discharged container does not have a label or corresponding label, in such case, removal of labels will take place. Label removal shall also take place when containers are returned for their resultization or in those cases when these do not correspond to IMDG code, for placing the correct label instead.

(n46). This service shall not be applicable for shipping, unloading or transhipment services for empty containers set forth in Section 1.1.2 and 1.1.3 of current Tariff Schedule.

SECTION 2	В	REAK BULK CARGO S	ERVICES								
Section 2.1	Standard Services	Nature	Unit of charge	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)
2.1.1	Standard Service to the Vessel - Services to the Vessel										
2.1.1.1	Use or rental of berth (Tariff per hour or fraction hour) (n1)	Regulated	Total LoA (m) x Hours	2.00	0% Rate			1.50	Tasa 0%		
2.1.2	Standard Service to Break Bulk Cargo										
2.1.2.1	Loading and Unloading of Break Bulk cargo (n47)	Regulated	Per Ton	2.50	0% Rate	17.78	3.20	2.50	0% Rate	17.70	3.19
(n47): Vessel port	ion of this tariff includes transference of the cargo from vessel to the quay, lashing/unlashing and tally. Payment of Vessel portion will be applicable to the shipping line or cargo consignee, upon or	anditions set forth in the ma	aritime shipping contract. If Ve	ssel portion of this tarif	f is charged to the ca	irgo consignee, then I	GV (IVA) shall be app	olicable to this tariff.			
Section 2.2	Special Services - Services to the Vessel (n6)	Nature	Unit of charge	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)
2.2.1	Hatches or Hold Lids (n7)										
2.2.1.1	Movement of ISO hatches	Regulated	Per move	0.00	0% Rate						
2.2.2	Re-stow of Break Bulk Cargo										

2.2.2.1	Re-stow via vessel	Regulated	Per ton	23.60	0% Rate			20.30	Tasa 0%		
2.2.2.2	Re-stow via quay (n48)	Negulated	Per ton	47.80	0% Rate			40.50	Tasa 0%		
2.2.3	Transhipment of Break Bulk cargo										
2.2.3.1	Transhipment of Break Bulk cargo - Complete cycle (n49)	Non Regulated	Per ton	30.10	0% Rate						
2.2.4	Special Cargo										
2.2.4.1	Loading/Unloading of project cargo without mobile harbor crane (n50)	Regulado	Por tonelada o m3			41.69	Tasa 0%			36.80	Tasa 0%
2.2.4.2	Loading/Unloading of project cargo with mobile harbor crane (n50)	regulado	Por tonelada o m3			58.37	Tasa 0%			49.80	Tasa 0%

n48): This is applicable to the cargo that is re-stowed in the same hold (level) or in another hold (level) of the vessel which in both cases cargo requires to be temporarily transferred to the quay

Break Bulk Cargo - Includes use of operational area until day 30 (n52)

2.4.1.3 Use of operational area from day 21 onwards

(neb). This a appricative to the catigud that is estimated in the same fund (every) of in advisor into (every) of int

Section 2.3	Special Service of Use of Operational Area (Port Terminal) - Services to the Cargo	Nature	Unit of charge	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)
2.3.1	Use of Operational Area - All traffic										
2.3.1.1	Days: 1 - 3 (Free time - included in Standard Service)	Regulated				Fre	96				
2.3.1.2	Days: 4 – 10 (Tariff for whole period or fraction period)		Per ton			26.00	4.68				
2.3.1.3	Days: 11 - 20 (Tariff per day or fraction day)	Non Regulated	Per ton/day			2.90	0.52				
2.3.1.4	Days: 21 - onwards (Tariff per day or fraction day)		Per ton/day			4.20	0.76				
Section 2.4	Special Service of Temporary Depot - Services to the Cargo	Nature	Unit of charge	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)
2.4.1	Temporary Depot Integrated Service - All traffic (n51)										
2.4.1.1	Break Bulk Cargo - Includes use of operational area until day 10		Per ton			31.70	5.71				
2.4.1.2	Use of operational area from days 11 – 20 (Price per day of day fraction)		Per ton/day			1.50	0.27				

See section 2.3.1.4 and 2.5.2 to 2.5.4 2.4.1.5 Other services (n53) (n51): This is applicable whenever APM Terminals Callao S.A. is appointed as Temporary Depot (code 3014) after cargo has been previously coordinated and accepted by APMTC (Subject to space availability). It includes the Standard Service cargo portion, use of operational area until day 10, temporary depot doc inte) and transmission to Customs. First three calendar days, which are free and are part of the Standard Service, are included within the 10 days (subsection 2.4.1.1) or 30 days (subsection 2.4.1.4) offered by the integrated temporary depot service for break-bulk cargo

Non Regulated

Per ton/day

(nSZ): Price of section 2.4.1.3 is applicable for use of operational area of temporary deposit of subsection 2.4.1.4 from day 31 and onwards.

(nSZ): For the other services, failfits and prices of sections from 2.5.2 to 2.5.4 are applicable independently from the number of storage days of break bulk cargo in the Terminal and regardless of the temporary deposit services chosen (either from section 2.4.1.1 or 2.4.1.4).

Section 2.5	Other Special Services to Break Bulk Cargo (Port Terminal & Temporary Depot) - Services to the Cargo	Nature	Unit of charge	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)
2.5.1	Use of Simple Depot - All traffic (n54)										
2.5.1.1	Days: 1 - 30 (Price per month or fraction month)	Non Regulated	Per m2			11.00	1.98				
2.5.2	Special Cargo										
2.5.2.1	Loading/Unloading of project cargo without mobile harbor crane (n50)		Per ton or m3			41.69	7.50			36.80	6.62
2.5.2.2	Loading/Unloading of project cargo with mobile harbor crane (n50)	Regulated	Per ton or m3			58.37	10.51			49.80	8.96
2.5.2.3	Special treatment to hazardous cargo using equipment and additional personnel - direct treatment (n55)	Regulateu	Per ton			22.28	4.01				
2.5.2.4	Special treatment to hazardous cargo using equipment and additional personnel - indirect treatment (n55)		Per ton			26.28	4.73				
2.5.3	Cargo Handling and horizontal movement operations										
2.5.3.1	Additional Movement as per request of the user or competent authority		Per ton			4.00	0.72				
2.5.3.2	Handling for registry of break bulk cargo (n56)		Per ton			4.00	0.72				
2.5.3.3	Deconsolidation/Consolidation of break bulk cargo with MAFIs (n57)	Non Regulated	Per MAFI			281.00	50.58				
2.5.3.4	Desconsolidation/Consolidation of Break Bulk cargo in MAFI's with Mobile Harbour Cranes (Price per shift or fraction shift) (n58)	Non Regulated	Per Shift			1,250.00	225.00				
2.5.3.5	Deconsolidation/Consolidation of MAFIs (n59)		Per MAFI	83.00	14.94						
2.5.3.6	Bagging or racking of goods in sacks or big bags (60)		Per ton			20.00	3.60				
2.5.4	Other services										
2.5.4.1	Additional weighing (This does not include extra movement)	Non Regulated	Per truck			21.00	3.78				

Non Regulated Per truck 21.00 3.78

(not): This service is explicable to receiving these Bulk corps, picting it in a designated area and storaging it for a period agreed by the exporter or importer and APM Terminals Callao S.A. This is applicable for Break Bulk. Cargo that will be finally shipped as another type of cargo (in containers mainly) or for another cargo that has been unloaded in container and is then storred as Break Bulk. Standard service shall not be pro-cargo in this case. I failt for this service is applicable as Simple Deport only. Space is subject to availability.

cargon man sase. same rorm servines a appraison eat Simple Upport only, Space a subject to availability.

(Indition Standard Services is basing and uniquation personance in a subject to More Inditional equipment and personnel are not to be provided, per Port Authority regulation or upon request of the user. Tariffs of sections 2.5.2.3 and 2.5.2.4 include Standard Service to break bulk cargo. If additional equipment and personnel are not to be provided, then Standard Service tariffs to break bulk cargo in the storage area.

(Indition This is replaced to septically the service in the se

(1960). This is applicable for assembly/place/ang or disassembly/separating of MAFIs for its delivery to the shipping line. This is invoiced to shipping lines.

(1960). Price applicable in case of solid bulk cargo racking on sacks or big bags. This price is also applicable when the load is transferred from a bag in poor condition onto a new bag. In both cases (bagging or racking) the service may be provided in the warehouse of the ship, dock or storage area. The minimum price to charge is USO 20.

Cargo (Tariff) Cargo (IGV)
44.20 7.96
(V)

Section 3.2	Special Services - Services to the Vessel	Nature	Unit of charge	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)
3.2.1	Re-stow of Ro-Ro cargo										
3.2.1.1	Re-stow via vessel	Regulated	Per ton	65.39	0% Rate			52.90	Tasa 0%		
3.2.1.2	Re-stow via quay (n62)	regulated	Per ton	121.02	0% Rate			104.00	Tasa 0%		
3.2.2	Transhipment of Ro-Ro cargo										
3.2.2.1	Transhipment of Ro-Ro cargo - Complete cycle (n63)	Non Regulated	Per ton	74.00	0% Rate						
3.2.3	Use of Operational Area - Transhipment (n64)										
3.2.3.1	Dias: 1-3 (Tiempo libre - included in the special service ro-ro cargo transhipment)			Fr	00						
3.2.3.2	Days: 4 – 10 (Price for total period or fraction period)	No Regulado	Per ton	48.80	Tasa 0%						
3.2.3.3	Days: 11 - onwards (Price per day or fraction day)		Per ton/day	20.00	Tasa 0%						

(n62): This is applicable to the cargo that is re-slowed in the same hold (level) or in another hold (level) of the vessel which in both cases cargo requires to be temporarily transferred to the quay

tion of cargo discharge operation of the vessel

(n64): Prices of this section are applicable to all types of vehicle and machinery.

3.3.1         Use of Operational Area - All Intellife (except transhipment) (n64)           3.3.1.1         Days: 1-3 (Free time - included in Standard Service)           3.3.1.2         Days: 4-10 (Price time - included in Standard Service)           Non Regulated         Per ton           90.00         16.20	Section 3.3	Special Service of Use of Operational Area (Port Terminal) - Services to the Cargo	Nature	Unit of charge	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)
3.3.1.2 Days: 4 – 10 (Price for total period or fraction period)  Non Remitated  Per ton  90.00  16.20	3.3.1	Use of Operational Area - All traffic (except transhipment) (n64)										
Non Requisited	3.3.1.1	Days: 1 - 3 (Free time - included in Standard Service)	Regulated				Fr	00				
	3.3.1.2	Days: 4 – 10 (Price for total period or fraction period)	Non Degulated	Per ton			90.00	16.20				
3.3.1.3 Days: 11 - onwards (Price per day or fraction day) 20.00 3.60	3.3.1.3	Days: 11 - onwards (Price per day or fraction day)	Non Regulated	Per ton/day			20.00	3.60				

Section 3.4	Special Service of Temporary Depot - Services to the Cargo	Nature	Unit of charge	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)
3.4.1	Temporary Depot Integrated Service - All traffic (n65)										
3.4.1.1	Ro-Ro cargo in general - Includes use of operational area until day 10		Per ton			121.40	21.85				
3.4.1.2	Use of operational area from day 11 onwards (Price per day or fraction day)	Non Regulated	Per ton/day			10.00	1.80				
3413	Other services (n66)		See section 3.5								

able wherever ATT Terminals Callao S.A. is appointed as Temporary Depot (code 3014) after cargo has been previously coordinated and accepted by APMITC (Subject to space availability). It includes the Standard Service cargo portion, use of operational area until day 10, temporary depot documentation emission (volante) and transmission to Customs. First three calendar days are free of charge and are part of the ole to all types of vet

(n66): Prices of section 3.5 apply to other services of Temporary Depot, independently from the number of storage days of RoRo cargo.

Section 3.5	Other Special Services to Ro-Ro Cargo (Port Terminal & Temporary Depot) - Services to the Cargo	Nature	Unit of charge	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)	Vessel (Tariff) Vessel (IG	V) Cargo (Tariff)	Cargo (IGV)
3.5.1	Handling Services in Storage area									
3.5.1.1	Handling for registry of Ro-Ro cargo (n67)	Non Regulated	Per ton			11.00	1.98			
3.5.1.2	Extra movement of Ro-Ro cargo (n68)	Non Regulated	Per ton			2.50	0.45			
3.5.2	Deconsolidation/Consolidation									
3.5.2.1	Deconsolidation/Consolidation of Ro-Ro cargo with MAFIs (n69)		Per MAFI			281.00	50.58			
3.5.2.2	Desconsolidation/Consolidation of Ro-Ro cargo en MAFI's with Mobile Harbour Cranes (Price per shift or fraction shift) (n70)	Non Regulated	Per Shift			1,250.00	225.00			
3.5.2.3	Deconsolidation/Consolidation of MAFI's (n59)		Per MAFI	83.00	14.94					
3.5.3	Vehicle Preparation for Delivery (PDI) Services									
3.5.3.1	Basic PDI service for light vehicles (n71) (n73)		Per unit			45.00	8.10			
3.5.3.2	Full PDI service for light vehicles (n71) (n73)		Per unit			70.00	12.60			
3.5.3.3	Basic PDI service for medium-sized trucks (n71) (n73)		Per unit			60.00	10.80			
3.5.3.4	Full PDI service for medium-sized trucks (n71) (n73)		Per unit			95.00	17.10			
3.5.3.5	Assemble/disassemble a vehicle part or accessory (accessorization)	Non Regulated	Per unit			18.00	3.24			
3.5.3.6	Ironing vehicle (price per piece or fraction of piece)	Non Regulated	Per Piece			29.50	5.31			
3.5.3.7	One-piece vehicle painting (price per piece or more than 1/2 piece)		Per Piece			52.00	9.36			
3.5.3.8	1/2-piece vehicle painting		Per 1/2 Piece			32.00	5.76			
3.5.3.9	Polished (price per piece or fraction of piece)		Per Piece			59.00	10.62			
3 5 3 10	Increase in weekly attention performance (n72)		Per unit			25.00	4 50			

(n67): This refers to handling of cargo using gangs and/or equipment for the registry or inspection of the cargo, as per request of the user, (cargo owner or consignee), or authorities.

(n88): Additional movement of Ro-Ro cargo within the Multipurpose North Terminal not included in the Standard Service, and upon request of the user, (cargo owner or consignee) or competent authority.

(n89): This is applicable for special treatment to Ro-Ro cargo in the storage area in order to separate/dispatch/unload or place/load RoRo cargo from metalic platforms (called MAFI trailers). This is invoiced to the cargo consignee

(7) Sentire includes the use of Mobile Harbour Cranes' availability.

The period of four (94) hours in order to separate withdraw/disphare and place/load Ro-Ro cargo from metallic platforms (Referred to as MAFI trailer). Price is invoiced to the Cargo Consignee, being applicable in those cases where Mobile Harbour Cranes are required for less than four-hour (4) period to Mobile Harbour Cranes' availability.

(n/1): The basic PDI service for light vehicles and medium trucks includes receiving the vehicle in the designated PDI area, a superficial and visual inspection for damage (dents from collisions), and the aesthetic conditioning of the vehicle's interior and exterior (which includes washing and vacuuming, dying, and applying plastic sheeling to various parts of the vehicle or medium truck). It also includes checking the fuel level and, if necessary, a fall supply of 3 gallons for light vehicles and agains for medium trucks. The full PDI services for light vehicles includes the basic service, but with a more thorough damage inspection, (covering stratches in the pairt and other exterior damage), as well as checking fluid levels, performing a mechanical inspection, and checking the electrical system's functionality and, if necessary, making electrical adjustments. For both the Basic and Full services, for five days of operational area use will be granted on the grant of the capture of operational area use will be granted on top of the aforementioned 10 days; that is, up to a total of 15 free days. These fee days of operational area use may be used before and/or after entity in the PDI area.

(n72) This applies if the user requires or requests that the weekly output be greater than 320 units

(n73): For the purposes of the PDI service, light vehicles are defined as established in Supreme Decree No. 25-2008-MTC, but in the case of the PDI, they must have a maximum weight of 6 tons per unit.

SECTION 4	SOLID BULK CARGO							
Section 4.1	Standard Service	Nature	Unit of charge	Vessel (Tariff) Vessel (IGV)	Cargo (Tariff)	Cargo (IGV) Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff) Cargo (IGV)

4.1.1	Standard Service to the Vessel - Services to the Vessel										
4.1.1.1	Use or rental of berth (Tariff per hour or fraction hour) (n1)	Regulated	Total LoA (m) x Hours	2.00	0% Rate			1.50	Tasa 0%		
4.1.2	Standard Service to Solid Bulk Cargo - Services to the Cargo										
4.1.2.1	Loading or Unloading of Solid Bulk	Regulated	Per ton			11.43	2.06			10.20	1.84
Section 4.2	Special Services - Services to the Vessel	Nature	Unit of charge	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)
4.2.1	Hatches or Hold Lids(n6)										
4.2.1.1	Movement of ISO hatches	Regulated	Per move	0.00	0% Rate						
Section 4.3	Special Services - Services to the Cargo		Unit of charge	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)
4.3.1	Remove/place of artificial separation in vessel's hold										
4.3.1.1	Remove of artificial separation in vessel's hold	Regulated	Per operation			2,000.00	360.00				
4.3.2	Use of Operational Area - All traffic										
4.3.2.1	Days: 1 - 5 (Free time - included in Standard Service) (n74)	Regulated				Fr	90				
4.3.2.2	Days: 6 – 15 (Price for whole period or fraction period)		Per ton			2.90	0.52				
4.3.2.3	Days: 16 – 30 (Price for whole period or fraction period)	Non Regulated	Per ton			4.00	0.72				
4.3.2.4	Days: 31 - onwards (Price per day or fraction day)		Per ton/day			0.40	0.07				
4.3.3	Other services										
4.3.3.1	Additional weighing (Extra movement is not included) (n75)	Non Regulated	Per truck			21.00	3.78				

Standard Service Nature Unit of charge Vessel (Tariff) Vessel (IGV) Cargo (Tariff) Cargo (IGV) Vessel (Tariff) Vessel (IGV) Cargo (Tariff) Cargo (IGV) 2.00

5.1.1.1 Use or rental of berth (Tariff per hour or fraction hour) (n1)
5.1.2 Standard Service to Liquid Bulk Cargo - Services to the Cargo
5.1.2.1 Loading or Unloading of Liquid Bulk Section 5.2 Special Services - Services to the Cargo Nature Unit of charge Vessel (Tariff) Vessel (IGV) Cargo (Tariff) Cargo (IGV) Vessel (Tariff) Vessel (IGV) Cargo (Tariff) Cargo (IGV) 5.2.1.1 Use of Commentor Barriers (Tariff per day or fraction day) (n76)
(n76) This is applicable for the use of contention barriers during loading and unloading operations of hazardous substances and/or goods such as hydrocarbons and other liquid bulk. Service includes placing and unpliadate in the maritime shipping contract. 2,089.00 376.02 Regulated Per day 2,089.00 376.02 1,841.00 ing contention barriers as well as cleaning and drying of contention barriers. Tariff of this service during loading lundeding operations of hazardous liquid bulk shall be applicable to the cargo consignee, unless other

PASSENGERS SECTION 6 Unit of charge | Vessel (Tariff) | Vessel (IGV) | Cargo (Tariff) | Cargo (IGV) | Vessel (Tariff) | Vessel (IGV) | Cargo (Tariff) | Cargo (IGV) 6.1.1 Standard Service to the Vessel - Services to the Vessel Use or rental of berth (Tariff per hour or fraction hour) (n1 Regulated Total LoA (m) x Hours 2 00 10.1.1.1 Use or remain or or remain or or remain or or reaction mourt (in) regulated (including the control of 23 67

SECTION 7 Section 7.1 Special Services - Services to the Vessel Nature Unit of charge Vessel (Tariff) Vessel (IGV) Cargo (Tariff) Cargo (IGV) Vessel (Tariff) Vessel (IGV) Cargo (Tariff) Cargo (IGV) 7.1.1 Use of Contention Barriers on Barriers (Tariff per hour or fraction hour) (n79) 2.089.00 376.02 1.841.00 331.38 Vessel Holds cleaning (price per gang-shift or fraction) (n81)
Unloading of non-hazardous solid waste (price per M3 or fraction) (n82)
Unloading of hazardous solid waste (price per M3 or fraction) (n83) Per gang/Shift Non Regulated 7.1.2.2 255.00 7.1.2.3 7.1.3 Per M3 400.00 72.00 Onlocating of nazaroous solid waste (price per M3 or fracts
Sludge Management
Oily Waste Management (n84)
Unloading of solid waste (price per M3 or fraction) (n85) 7.1.3.1 7.1.3.2 Per truck 1.326.00 238.68 Per M3 357.00 64.26 Non Regulated 7133 Management of non-Hazardous solid waste - non recoverables (price per M3 o fracción) (n86) Management of Hazardous solid waste (price per M3 or fraction) (n87) 357 00 64 26 7.1.3.4 Per M3 500.00 90.00 7.1.4 7.1.4.1 Water supply (n88) Non Regulated Per ton 12.00 2.16 7 1 4 2 Phone serv

(178) Services applicable to all type of vessels.
(179) This is applicable for the use of contention barriers during fuel supply operations to the vessel. Service includes placing and unplacing contention barriers as well as cleaning and drying service. Tariff for contention barriers shall be applicable to the shipping line or its representative.

(n75): This service shall be provided within facilities of Multipurpose North Terminal as per request of user or authorities. (Extra) movement of solid bulk cargo is requested from its stacking area to the Gales area. Such service does not include extra movement service

(n81): This service refers to picking up solid wastage or obsert the vessel such as materials from ashing unlasting unlasting

(not) In service resers to pooning up soon westage nonceaut the vessel across a materias to manifestal to make (evaluage of bits) and included cargo, etc.) not classified as dangerous. Service includes temporary store in the terminal facility if necessary, as well as final wested disposal. Minimum invoice shall be USD 255.
(n83): This refers to unloading hazardous solid westage contended reversel from the dearning of the ship's holds and final disposal of wastes from vessel. Minimum invoice shall be USD 400.
(n84): Pice of this service includes collection and final disposal of waste from vessel.
(n84): Pice of this service includes collection and final disposal of waste from vessel.
(n85): This service refers to discharge on collecting from vessels recoverable waste from messels (such as; plastics or metal cords, wood, cardboard, paper, magazines and any other of the same type, classified as non-diangerous. Price of service applies to those cases where APM Terminals Callao S.A. does not carry out vessel hold cleaning service (section 7.1.2). Minimum invoice shall be USD 357.
(n85): This service refers to discharging or collecting from vessels incurred to the case of the control of the cases where APM Terminals Callao S.A. does not carry out vessel hold cleaning service (section 7.1.2). Minimum invoice shall be USD 357.
(n87): This service refers to discharging or collecting from vessels house cases where APM Terminals Callao S.A. does not carry out vessel hold dearing service (section 7.1.2). Minimum invoice shall be USD 357.

(n88): This shall be provided according to the Terminal's berth/pier availability or capacity and shall be charged to the shipping line or representative.

SECTION 8	COMMERCIAL POLICIES (n89)												
Section 8.1	Cancellation of loading	Nature	Unit of charge	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)		
8.1.1	Cancellation of container loading and withdraw of container (n90)												
8.1.1.1	20' Container	Non Regulated	Per container		220.00 320.00		22		39.60				
8.1.1.2	40' Container	Non Regulated	Per container				57.60						
8.1.2	Cancellation of break bulk and ro-ro cargo loading (n91)												
Section 8.2	Berthing window reservation	Nature	Unit of charge	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)	Vessel (Tariff)	Vessel (IGV)	Cargo (Tariff)	Cargo (IGV)		
8.2.1	Contracting of berth window reservation for vessels												
8.2.1.1	Contracting of guaranteed berth window reservation for regular traffic vessels with ro-ro cargo	Non Regulated	by vessel landfall	10,000.00	1,800.00								
(n89): For further of	tetaits on the scope and conditions of applicable discounts, offers and promotions, see Charter V of Tariffs Regulation and Commercial Policy of APM Terminals Callao S.A. published on website:	www.apmterminalscallao.d	om.pe										

[60]: This is applicable wherever the user (consignee, representative or shipping line) withdraws a container (withdraws a container (withdraws a through inter-terminal gate applies for full and empty containers. In the case of transhipment containers the prices in this section apply only to those containers that enter through the inter-terminal gate but cancel the loading through the North Terminal and leave again through the inter-terminal gate. Tariffs of section 1.2 (for empty) or 1.3 (for full) that refer to the use of the operational area applied if the container stays more than 48 hours.

tion) shall be charged as well as the use of operating area (Port Terminal). For use of operating area, the tariffs stated in section 2.3.1 and 3.3.1 shall be applicable if break bulk and ro-ro cargo

SECTION 9	SURCHARGES (n/2)									
Section 9.1	General surcharges	Unit of charge	USD (\$)	IGV	Total					
9.1.1	Certificates and receipts									
9.1.1.1	New receipt printing	Per receipt	20.00	3.60	23.60					
9.1.1.2	Issue of weight certificate	Per document	5.00	0.90	5.90					
9.1.1.3	Issue of copy of weighing ticket and/or tally note	Per document	5.00	0.90	5.90					
9.1.1.4	Letter of amendment and/or sending new details to SUNAT	Per document or new transmission	28.00	5.04	33.04					
9.1.1.5	New collection settlement (n93)	Per document	8.00	1.44	9.44					
9.1.2	Returned checks and default (n94)									
(n92): For further	information and scope, refer to Charter V, section 5.6 of the Tariffs Regulation and Commercial Policy of APM Terminals Callao S.A. published on website: www.apmterminalscallao.com.pe									

(n94): Surcharge not subject to IGV. See section 5.6 f the Tariffs Regulation and Commercial Policy of APM Terminals Callao S.A.

Section 9.2	Surcharges applicable to the Vessel	Unit of charge	USD (\$)	IGV	Total
9.2.1	Surcharges for non-use or non-release				
9.2.1.1	Offset for not used crew (n95)	Per crew/hour	162.00	29.16	191.16
9.2.1.2	No release of berth	Per hour	600.00	Not subject to IGV	600.00
9.2.2	Liquid Spill Management Surcharges				
9.2.2.1	Liquid spill management (n96)	Per case	Subj	ect to price quote	

(n96): This acolies when liquids are soiled on the water area or quay of the Multipurpose North Terminal due to user's responsibility. It includes removing and/or collecting and final disposal of the substances, in accordance with current regulations. This surcharge will acoly only when APMT carries out the soil

Section 9.3	Surcharges applicable to the Cargo	Unit of charge	USD (\$)		Total
9.3.1	Change of status				
9.3.1.1	Change of container status	Per container	50.00	9.00	59.00
9.3.1.2	Change of non-containerized cargo status	Per B/L	108.60	19.55	128.15
9.3.2	Late arrival				
9.3.2.1	Late arrival of containers	Per container	200.00	36.00	236.00
9.3.2.2	Late arrival of break bulk cargo	Per ton	2.40	0.43	2.83
9.3.2.3	Late arrival of Ro-ro cargo	Per unit	150.00	27.00	177.00
9.3.2.4	Late arrival of solid bulk cargo	Per unit	1.50	0.27	1.77
9.3.2.5	Late arrival of liquid bulk cargo (n97)	Per unit	1.00	0.18	1.18
9.3.3	Late submission				

9.3.3.1	Late submission of shipping documents of containers (n98)	Per DAM	85.00	15.30	100.30				
9.3.4	Missing, cancelling or rescheduling appointments (n99)								
9.3.4.1	Missing an appointment (n100)	Per appointment	16.00	2.88	18.88				
2.3.4.2	Cancelling or rescheduling an appointment (n101)	Per appointment	10.50	1.89	12.39				
9.3.5	Rescheduling services								
9.3.5.1	Rescheduling movement for containers (n102)	Per container	100.00	18.00	118.00				
9.3.6	Other surcharges to the Cargo								
9.3.6.1	Bulk cargo re-delivery to truck (n103)	Per truck	225.00	40.50	265.50				
9.3.6.2	Liquid spill management (n96)	Per case	Per case Por cotización						
	(n97): This applies only to shipping through tanker trucks.								
(n89): This applies only to the document submission of containers that are taken into the Temporary Depot (3014) out of the cut off set by APM TERMINALS CALLAO S.A.									
(n99): Appointmen	(n99): Appointments will lest one (01) hour.								
(n100): Missing a	100). Missing an appointment mapper, user door, not attend within the pooled graphed by ADM TEDMINALS CALLAGEA, and also whose the user from the appointment during the pooled behavior, not be provided by ADM TEDMINALS CALLAGEA, and also whose the user from the appointment during the pooled behavior on the provided provided by ADM TEDMINALS CALLAGEA, and also whose the user from the appointment during the pooled behavior on the provided provided by ADM TEDMINALS CALLAGEA.								

(n101): The cancellation surcharge applies when the user cancels the appointment within eight and one hours before the start of the appointment. The rescheduling surcharge applies when the user reschedules the appointment within eight and one hours before the start of the first appointment. The rescheduling surcharge applies when the user reschedules the appointment within eight and one hours before the start of the first appointment.

[n102]: The surcharge applies when the customer or its representative, after failing to attend or cancelling a scheduled and approved container movement service (related to an inspection, consolidation deconsolidation or other service), requests the rescheduling of such service. The user requests rescheduling when he does not show up for the service within the scheduled time or when he cancels the service outside the deadlines established in the Operating Regulations. Note that, according to the Operations Regulations, the deadline to cancel a mobilization service is before 6 p.m. on the same day the mobilization request is submitted. This surcharge applies for each escheduling requested (if the user request two reschedulings for the same container the user will be charged twice the surcharge). Examples of services that require container mobilization: customs inspection (by red channel), SCNASA BOE, loading continuent mention, consolidation/deconsolidation, among others.

[n103]: This applies when solid cargo falls to the ground while uploading it from silos to truck or from hopper to truck use to driver's or truck's inefficiency (e.g. not dosing truck's door properly). It includes slab cleaning and bulk cargo re-delivery to truck. Surcharge will be charged to the user (consignee, customs agent, or others)

### COMMERCIAL POLICIES

### DISCOUNTS, OFFERS AND PROMOTIONS

Discounts, offers and promotions are commonly accepted practices that answer to compensatory circumstances which shall be determined by APM TERMINALS CALLAO S.A. in compliance with the principle of non-discrimination, equal in this regard, APM TERMINALS CALLAO S.A. shall generally establish the following commercial practices of discounts, offers and promotions:

Discounts
Discounts shall be effective during the period established by APM TERMINALS CALLAO S.A. User shall consider the possibility that discounts may not be renewed.

Promotions and/or temporary offers

APM TERMINALS CALLAO S.A. may temporarily provide, as part of its commercial policy, offers and/or promotions, as long as permanent equivalent con-

For cases forseen in Section 5.1.1 and 5.1.2 of Tariffs Regulation and Commercial Policy, APM TERMINALS CALLAD S.A. shall publish on its website the effective term for temporary discounts, promotions and/or offices with details for the advnowledgement of users.

APM TERMINALS CALLAD S.A. shall also publish termination date or modification of discounts, promotion and/or offices on its website on time.

CRITERIA POR THE APPLICATION OF DISCOUNTS, OFFERS AND PROMOTIONS
APM TERMINALS CALLAO S.A. shall provide discounts, offers and/or promotions to fees of current Tariffs and/or Prices applicable to the Vessel and/or the Cargo considering the following criteria:

Volumentary Companies Comp

Generation for more demand of services provided to Vessels, Cargo and others.

Generation of more productivity,

in the case of Services provided in Vessels discounts, offers and/or promotions shall be established according to docking frequency, cargo volume and other oriteria that are applicable by its in the event of Services provided to the Cargo, discounts, director promotions shall be established according to cargo volume and other oriteria that are applicable by its nature.

Secondary, offers and/or promotions are addressed as It ally improprised and formally represented User.

### SPECIAL SERVICES WITH DISCOUNTS FIXED IN THE CONCESSION AGREEMENT

NNLS CALLAO S.A. shall provide the following Special Services established in the Concess Hatch movilization (ISO Hatches) Re-stowage of containers (Hold or via quey) Shippingulroiding and delivery no ISO/OOG (Additional) - container 20 and 40 foot ion Contract with a hundred per cent discount (100%), whenever the User requests this as complementary service to the Standard Service

Shipping juricologing and delivery no ISOLOUG (Additional) - conta Horizontal movement in the Multipropose North Terman Handling of Cargo per registry (with or without forklifts) Reefer energy Reefer inspection and monitoring Assembling and disassembling of clip system on reefer Gensets

These Special Services shall be provided considering principles set forth in the Tariff Regulation and Commercial Policy, and the Concession Contract, as APM TERMINALS CALLAD S.A. and the User may not bridge rules and practices that may after competition of Terminal Services, or that implies abusive and inefficient that may after competition of Terminal Services, or that implies abusive and inefficient that may be a support of the services and inequality of the services and the services are required in the services and inequality of the services are required in the services are required in the services and in the services are required in the serv

Description details of each service are included in Chapter VII of Services Summary of the Tariff's Regulation and Commercial Policy.

### RESERVE OF GUARANTEED BERTHING WINDOW

Space may be reserved for Vessel's docking provided that this has been previously arranged between APM TERMINALS CALLAO S.A. and the shipping lines as per terms and conditions set forth in the Operations Regulations and procedures from the Berthing Window Res Clause 8.14 of the Concession Contract.

### APPLICATION OF TAX ON GENERAL SALES (IGV)

The rates and prices of standard and special services taxed at zero percent (0%) rate, which by version 4.0 of Tariff Scheme were listed as exempt from VAT, refer to those services according to the VAT Law considered as exports of services. Rate of 0% will apply as long as the User is considered under the category "Inte-Cargo Carrier", Other than that the current rate of VAT amounting to 18% will apply.

PAYMENT CONDITIONS

Currency of Payment: Tariff and prices in American Dolars (US\$) may be cancelled in Nuevos Soles (St.) according to the average sale exchange rate of the financial system indicated by Superintendence of Bank, Insurance and AFP (SBS) at the date of payment.

Form of Payment: Payments may be through bank account transference, deposit or by check in the same bank that for this purpose will be assigned by APM TERMINALS CALLAO S.A.

Date of Payment: Payments shall be done according to the following considerations:

O Special Services - Export: A the income about the Contract Contr

# OFFERS AND REDUCTIONS OF TARIFFS AND PRICES

Offices or tainfils and/or prices reductions apply to individual companies, but also to economic groups in general or associations of companies that import resefer containers. In order for the offer or discount to apply to the companies forming an economic group, the following conditions must be met, at least; (i) that the principal or parent company is the orne requesting the offer; and (ii) that the principal or parent company has sent the list of companies with the intended or will access to the offer, this is in must be submitted by present of a settler in which it is supported, in accordance with national law, that the companies on the said list are part of its economic group; the terms of which company or companies will be invoiced will be coordinated with the main or parent company. For the promotion or discount to apply to the companies that make up an association of reseter contains imported, in accordance must be met, at least; (ii) that the principal or parent company is the orne requesting the offer. The principal or parent company is the orne requesting the orne requestion or the orne requesting

all cases of the offers isleed below, the latfill or price to be charged for storage a see completed days off of use of the operating area that each promotion grants are not retroactive. For example, item ii) of offer 1.2 grants a latd of 30 fee days; if the user remains in the storage area until day 35, the price of subsection 1.3.1.7 of the Tariff List will apply for the period from day 31 to day 35. Another example, item ii) of offer 1.2 grants a latd of 10 fee days; if the user's cargo or container remains in the storage area until day 34, then the user will be charged the price of subsection 1.3.2.7 of the Tariff List for the range from day 21 to day 28, and the price of subsection 1.3.2.7 of the Tariff List for the range from day 21 to day 34.

# (Group 1) SPECIAL OFFER FOR DRY CARGO CONTAINERS IN APM TERMINALS CALLAO'S TEMPORARY DEPOT

APIX Terminals Calabo (APIXTC) has set special offers for the containers processed as Temporary (apost service. Each promotion is independent (not cumulative); for example, if the user doscess promotion 1.1, they would not be able to apply to promotion 1.2, they remotion as group composition 1.1 or promotion 2.1, and 1.0. The Promotions are grouped by the following undergood by the following unde

# 1.1 Discount on Integrated Service Price and extension of free days – dry cargo import

- Target group: Importers of dry cargo containers (consignees or its representatives -customs brokers, logistics operators, etc.).
- Segment: Importers who address dry cargo to APMTC's Temporary Depot.
- rators, etc.) who guarantee minimum amounts equal to or greater than 200, 300, 800, 1.6 thousands and 6 thousand containers through the APMTC's Temporary Depot will be granted discounts and extensi Temporary offer: Importers of dry cargo (consignees or its representative Temporary Depot, sections 1.4.1.1 and 1.4.1.2 of the Tariff List) as below:

  - (ii) > 300 to 799 containers = USD 45 discount from the price set out in sections 1.4.1.1 and 1.4.1.2 of the Tariff (reaching a total of 20 free days)
  - (ii) 2 800 to 1,599 containers = USD 45 discount from the price set out in sections 1.4.1.1 and 1.4.1.2 of the Tariff List for import containers and 20 extra free days to those already granted in subsections 1.4.1.1 and 1.4.1.2 of the Tariff (reaching a total of 30 free days) (iv) ≥ 1,800 to 5,999 containers = USD 55 discount from the price set out in sections 1.4.1.1 and 1.4.1.2 of the Tariff Lest for import containers and 20 extra free days to those already granted in subsections 1.4.1.1 and 1.4.1.2 of the Tariff (reaching a total of 30 free days)
  - (v) ≥ 6,000 containers = USD 65 discount from the price set out in sections 1.4.1.1 and 1.4.1.2 of the Tariff List for import containers and 20 extra free days to those already granted in subsections 1.4.1.1 and 1.4.1.2 of the Tariff (reaching a total of 30 free days)
- Effective: Importers may apply or request this promotion from February 26, 2025 to December 31, 2025. Users who had promotional agreements prior to February 26, 2025, will be governed by the minimum volumes (ranges i, ii, ii, iv, and v) and conditions outlined in version 13.9 or earlier of the Tariff List.

# - Requirements:

The minimum volumes of the offer will be accounted for annually. That is, in the 12 months following the date of the agreement (between APMITC and the user) to start accounting for the volumes (for example and as a hypothetical case; if the user's request for offer is accepted and it is agreed to start accounting on February 1, 2024, then the accounting will be made until January 31, 2025). For this purpose, users must submit to the commercial area of APMITC the projections of containers to be imported in the annual period indicated above, in order to determine the availability of space at the terminal.

APMITC will carry a review every three (IGI) months of the imported volume. If it is observed that in the responsed quarter at least 45% of the committed annual volume has not been met, the Offer will be cancelled; and if the review of the third quarter finds that at least 70% of the required annual volume has not been met, the Offer will be cancelled. This consideration applies to offers whose movement accounting starts on or after January 1, 2024.

For offers already agreed before January 1, 2024, the penalties described in version 12.5 or earlier of the Tariff List apply.

Acceptance to any of the offers ranges (whether i, ii, iii, iv or v) will be subject to an assessment of the projections and/or that the user has mobifized by the Port of Calao, in the year preceding the agreement to start accounting, at least 80% of the volume common than the common transfer of the projections and or that the user has mobifized by the Port of Calao, in the year preceding the agreement to start accounting, at least 80% of the volume common transfer or the projections are not provided by the Port of Calao, in the year preceding the agreement to start accounting, at least 80% of the volume common transfer or the projection and the projection are not provided by the Port of Calao, in the year preceding the agreement to start accounting, at least 80% of the volume common transfer or the projection and the projection and the projection are not provided by the Port of Calao, in the year preceding the agreement to start accounting, at least 80% of the volume common transfer or the projection are not provided by the Port of Calao, in the year projection are not provided by the Port of Calao, in the year projection are not provided by the Port of Calao, in the year projection are not provided by the Port of Calao, in the year projection are not provided by the Port of Calao, in the year projection are not provided by the Port of Calao, in the year projection are not provided by the Port of Calao, in the year projection are not provided by the Port of Calao, in the year projection are not provided by the Port of Calao, in the year projection are not provided by the Port of Calao, in the year projection are not provided by the Port of Calao, in the year projection are not provided by the Port of Calao, in the year projection are not provided by the Port of Calao, in the year projection are not provided by the Port of Calao, in the year projection are not p

In the event that the offer has been cancelled to a user and the user requests access to a new offer (whether i, ii, ii, iv or v), the new access will be granted if the following condition is met: that during the period of the cancelled offer (due to failure to comply with the volume commitment beth DT 3014 equal to or more than 80% of the total containers that have been directed through a temporary deposit, (0014 or extraport) and unbaded/loaded by North Multi-purpose Terminal.

In case of chances in the prices of the subsections 1.4.1.1 o 1.4.1.2 of the Tariff List. the discounts of USD 45, USD 55 or USD 65 apoly to these modified prices in force at the time of invoicing. See invoicing details in the Regulation of Tariffs and Commercial Policy of APM Terminals Callac - This offer is subject to availability of space determined by APMTC.

# unt on Integrated Service Price and extension of free days - dry cargo export

- Target group: General users (exporters, forwarders or its representatives -oustoms brokers, logistics operators, etc- and other users) of containers with dry export cargo.
- Segment: Exporters who address dry cargo to APMTC's Temporary Depot.
- orary offer. Exporters (browarders or its representatives -customs brokers, logistics operators, etc.) who guarantee minimum amounts of more than 300, 700, 800, and 6000 containers using the AFMTC's Temporary Depot will be granted discounts and extensions of the operating area use (taking into account the 7 days including in the Integrated Service of Export Temporary Depot vertices 1.4.2.1 and 1.4.2.2. of the Tariff List), according to the table below
  - (i) ≥ 300 to 699 containers = 3 extra free days (resulting in a total of 10 free days)
  - (i) ≥ 700 hasta 799 contenedores = USD 20 discount from the price set out in sections 1.4.2.1 and 1.4.2.2 of the Tariff List for export containers and 3 extra free days (resulting in a total of 10 free days)
  - (iii) ≥ 800 to 5,999 containers = USD 20 discount from the price set out in sections 1.4.2.1 and 1.4.2.2 of the Tariff List for export containers and 8 extra free days (resulting in a total of 15 free days)
  - (iv) ≥ 6,000 containers = USD 20 discount from the price set out in sections 1.4.2.1 and 1.4.2.2 of the Tariff List for export containers and 13 extra free days (resulting in a total of 20 free days)
- Effective: Importers may apply or request this promotion from February 26, 2025 to December 31, 2025. Users who had promotional agreements prior to February 26, 2025, will be governed by the minimum volumes (ranges i, ii, iii, iv, and v) and conditions cultined in version 13.9 or earlier of the Tariff List.

The minimum volumes of the offer will be accounted for annually. That is, in the 12 months following the date of the agreement (between APMITC and the user) to start accounting for the volumes (for example and as a hypothetical case; if the user's request for offer is accepted and it is agreed to start accounting on February 1, 2024, then the accounting will be made until January 31, 2025). For this purpose, users must submit to the commercial area of APMITC the projections of containers to be imported in the annual period indicated above, in order to determine the availability of space at the terminal. APMITC will carry a review every three ((3) months of the exported volume. If it is observed that in the first quarter at least 45% of the committed annual volume has not been met, the Offer will be cancelled; and if the review of the third quarter finds that at least 70% of the required annual volume has not been met, the Offer will be cancelled. This consideration applies to offers whose movement accounting starts on or after January 1, 2024.

For offers already agreed before January 1, 2024, the penalties described in version 12.5 or earlier of the Tariff List apply.

splance to any of the offers ranges (whether i, ii, iii or iv) will be subject to an assessment of the projections and/or that the user has mobilized by the Port of Callao, in the year preceding the agreement to start accounting, at least 80% of the volume committed

In the event that the offer has been cancelled to a user and the user requests access to a new offer (whether i. ii. ii. or iv), the new access will be granted if the following condition is met: that during the period of the cancelled offer (due to failure to comply with the v DT 3014 equal to or more than 80% of the total containers that have been directed through a temporary depost (3014 or extraport) and unbased/loaded by North Multi-purpose Terminal.

In case of changes in the prices of the subsections 1.4.2.1 or 1.4.2.2 of the Tariff List. the discounts of USD 20 apply to these modified prices in force at the time of invoicing. See invoicing details in the Regulation of Tariffs and Commercial Policy of APM Terminals Callac

### ount on Integrated Service Price and extension of free days - Import/Export of dry cargo

- Target group: General users (importers/consignees and exporters/forwarders or its representatives -customs brokers, logistics operators, etc.- and other users) of containers with import/export dry cargo. If import containers only are mobilized, the discount on item 1.1 will be granted: and if export containers only are mobilized. The discount on item 1.2 will be granted.

Segment: Users who address dry cargo to APMTC's Temporary Depot.

- Temporary offer: Users who mobilize import and export dry cargo containers logether who guarantee minimum amounts equal or more than 300, 700, 800, 1600 and 6,000 containers through the APMITC's Temporary Depot will be granted discounts and extensions in the use of operational area with respect to the prices and days of use of free operating area considering in the Special Integrated Temporary Depot and Export Dry Containers (Sections 1.4.1.1, 1.4.1.2, 1.4.2.1 and 1.4.2.2 of the Tariff List), according to the table below.

- (ii) ≥ 300 to 699 containers = 10 additional free days with respect to the free days of use of operational area indicated in sections 1.4.1.1 and 1.4.1.2 of the Tariff List for import containers (which results in a total of 20 free days) and 3 additional free days of the days of t
- (i) 2 700 to 799 containers = USD 45 discount and 10 additional free days with respect to the prices and fee days indicated in sections 1.4.21 and 1.4.2.2 of the Tariff List for export containers (resulting in a total of 10 fee days) and USD 20 discount and 3 additional free days with respect to the prices and days indicated in sections 1.4.21 and 1.4.2.2 of the Tariff List for export containers (resulting in a total of 10 fee days)
- (ii)  $\geq$  800 to 1,599 containers = USD 45 discount and 20 additional free days with respect to the prices and days indicated in sections 1.4.2.1 and 1.4.2.2 of the Tariff List for import containers (resulting in a total of 30 fee days) and USD 20 discount and 8 additional free days with respect to the prices and days indicated in sections 1.4.2.1 and 1.4.2.2 of the Tariff List for export containers (resulting in a total of 15 free days)
- (iv) > 1.600 to 5.999 metainant = 1150 55 discrenat and 20 additional flow days with research to the nerices and flow indirected in sections ( 4.2.1 and 1.4.2.2 of the Tariff list for innort metainant (resulting in a hold of 10 feet days with research to the nerices and days indirected in sections ( 4.2.1 and 1.4.2.2 of the Tariff list for a rs (resulting in a total of 15 free days)
- (y)  $\geq$  6,000 to 11,999 containers = USD 65 discount and 20 additional free days with respect to the prices and days indicated in sections 1.4.1 and 1.4.1.2 of the Tariff List for import containers (resulting in a total of 30 free days) and USD 20 discount and 13 additional free days with respect to the prices and days indicated in sections 1.4.2.1 and 1.4.2.2 of the Tariff List for import containers (resulting in a total of 30 free days)
- (i) ≥ 1/2/00 containers = USD \$20 decount and 20 additional free days with respect to the prices and free days indicated in sections 1.4.1.1 and 1.4.1.2 of the Tariff List for import containers (resulting in a total of 30 free days) and USD 20 decount and 13 additional free days with respect to the prices and days indicated in sections 1.4.2.1 and 1.4.2.2 of the Tariff List for import containers (resulting in a total of 30 free days) and USD 20 decount and 13 additional free days with respect to the prices and days indicated in sections 1.4.2.1 and 1.4.2.2 of the Tariff List for import containers (resulting in a total of 20 free days) and USD 20 decount and 13 additional free days with respect to the prices and days indicated in sections 1.4.2.1 and 1.4.2.2 of the Tariff List for import containers (resulting in a total of 30 free days) and USD 20 decount and 13 additional free days with respect to the prices and days indicated in sections 1.4.2.1 and 1.4.2.2 of the Tariff List for import containers (resulting in a total of 30 free days) and USD 20 decount and 13 additional free days with respect to the prices and days indicated in sections 1.4.2.1 and 1.4.2.2 of the Tariff List for import containers (resulting in a total of 30 free days) and USD 20 decount and 13 additional free days with respect to the prices and days indicated in sections 1.4.2.1 and 1.4.2.2 of the Tariff List for import containers (resulting in a total of 30 free days) and USD 20 decount and 13 additional free days with respect to the prices and the section 1.4.2.2 of the Tariff List for import containers (resulting in a total of 30 free days) and USD 20 decount and 13 additional free days with respect to the prices and the section 1.4.2.2 of the Tariff List for import containers (resulting in a total of 30 free days) and USD 20 decount and 20 additional free days with respect to the prices and the section 1.4.2.2 of the Tariff List for import containers (resulting in a total of 30 free days) and USD 20 decount and 20 additional free days wi

- Effective: Importers may apply or request this promotion from February 26, 2025 to December 31, 2025. Users who had promotional agreements prior to February 26, 2025, will be governed by the minimum volumes (ranges i, ii, ii, iv, and v) and conditions outlined in version 13.9 or earlier of the Tariff List.

The minimum volumes of the offers i) at v) will be accounted for annually, that is to say, in the 12 months following the date of the agreement (between APMIC and the user) to start counting the volumes (for example and as a hypothetical case; if the user's request for offer is accepted and it is agreed that the positing on February 1, 2025, he the accounting will be made until January 31, 2025, in the case of Promotion vi, the accounting is also annual and the beginning of the velocity of this offer is June 27, 2025, but the accounting of also annual period indicated above, in order than 1, 2025. For this purpose, users must submit to the commercial area of APMIC the projections of containers that will be imported and exported in the annual period indicated above, in order than 1, 2025. For this purpose, users must submit to the commercial area of APMIC the projections of containers that will be imported and exported in the annual period indicated above, in order than 1, 2025. For this purpose, users must submit to the commercial area of APMIC the projections of containers that will be imported and exported in the annual period indicated above, in order than 1, 2025. For this purpose, users must submit to the commercial area of APMIC the projections of containers that will be imported and exported in the annual period indicated above, in order to account the commercial area of APMIC the projection of containers that will be imported and exported in the annual period indicated above, in order to account the commercial area of APMIC the projection of containers that will be imported and exported in the case of APMIC the projection of containers that will be imported and exported in the case of APMIC the projection of containers that will be imported and exported in the case of APMIC the projection of the account that the case of APMIC the projection of the account that the case of APMIC the projection of the account that the case of APMIC the projection of the account that the case of APMIC the projection of the

APMITC will carry a review every three ((3) months of the imported/exported volume. If it is observed that in the first quarter at least 20% of the committed annual volume has not been met, the Offer will be cancelled, and if the review of the third quarter finds that at least 70% of the required annual volume has not been met, the Offer will be cancelled. This consideration applies to offers whose movement accounting starts on or after January 1, 2024.

For offers already agreed before January 1, 2024, the penalties described in version 12.5 or earlier of the Tariff List apply.

Acceptance to any of the offers ranges (whether i, ii, ii, iv, or vi) will be subject to an assessment of the projections and/or that the user has mobilized by the Port of Calso, in the year proceeding the agreement to start accounting, at least 80% of the volume committed.

In the event that the offer has been cancelled to a user and the user requests access to a new offer (whether i, ii. ii, or v), the new access will be granted if the following condition is met: that during the period of the cancelled offer (due to failure to comply with the volume commitment between the accounting start date and the cancellation date) it is verified that this user has handled directed by DT 3014 equal to or more than 80% of the total containers that have been directed through a temporary deposit (2014 or extraport) and unbased/loaded by North Multi-purpose Terminal.

In case of changes in the prices of the subsections 1.4.1 or 1.4.2 of the Tariff List, the discounts apply to these modified prices in force at the time of invoicing. See invoicing details in the Regulation of Tariffs and Commercial Policy of APM Terminals Callao

### 1.4 Discount on the price of integrated service and extension of free days - Import/Export of dry cargo (only for new customers and without volume commitment)

- Segment: Users who address dry cargo to APMTC's Temporary Depot.

- Temporary offer: Users who mobilize import and export dry cargo containers who guarantee minimum amounts equal or more than 150, 400, 800 and 1,500 containers through the APMTC's Temporary Depot will be granted a discount on the price of the Special Integrated Temporary Depot Service for Import and Export Dry Containers (Sections 1.4.1.1, 1.4.1.2, 1.4.2.1 and 1.4.2.2 of the Tariff List) and an exist of days free of use of operational area, according to the table below:

(i) ≥ 150 to 399 containers = For import containers a discount of USD 45 on the price indicated in sections 1.4.1.1 and 1.4.1.2 of the Tariff List preaching a total of 20 fee days); while for export containers a discount of USD 20 on the price indicated in sections 1.4.2.1 and 1.4.2 of the Tariff List and 3 free days in addition to those already granted in sections 1.4.1.1 and 1.4.1.2 of the Tariff List and 3 free days in addition to those already granted in sections 1.4.2.1 and 1.4.2.2 of the Tariff List preaching a total of 10 free days).

(i) ≥ 400 to 799 containers = For import containers a discount of USD 45 on the price indicated in sections 1.4.1.1 and 1.4.1.2 of the Tariff List and 20 free days in addition to those already granted in sections 1.4.1.1 and 1.4.1.2 of the Tariff List (reaching a total of 30 free days); while for export cont. Tariff List and 50 free days in addition to those already granted in sections 1.4.1.1 and 1.4.1.2 of the Tariff List (reaching a total of 15 free days).

(iii > 800 to 1,499 containers = For import containers a discount of USD 55 on the price indicated in sections 1.4.1 and 1.4.1.2 of the Tariff List (reaching a total of 30 fee days), while for export containers a discount of USD 20 on the price indicated in sections 1.4.2.1 and 1.4.2.2 of the Tariff List (reaching a total of 1.5 fee days) and office of the Tariff List (reaching a total of 1.5 fee days).

(ii) ≥ 1,500 containers = For import containers a discount of USO 65 on the price indicated in sections 1.4.1 and 1.4.1.2 of the Tariff List (reaching a total of 30 free days); while for export containers a discount of USO 20 on the price indicated in sections 1.4.2 and 1.4.2.2 of the Tariff List (reaching a total of 30 free days); while for export containers a discount of USO 65 on the price indicated in sections 1.4.2.1 and 1.4.2.2 of the Tariff List (reaching a total of 30 free days); while for export containers a discount of USO 20 on the price indicated in sections 1.4.2.1 and 1.4.2.2 of the Tariff List (reaching a total of 30 free days); while for export containers a discount of USO 65 on the price indicated in sections 1.4.2.1 and 1.4.2.2 of the Tariff List (reaching a total of 30 free days); while for export containers a discount of USO 65 on the price indicated in sections 1.4.2.1 and 1.4.2.2 of the Tariff List (reaching a total of 30 free days); while for export containers a discount of USO 65 on the price indicated in sections 1.4.2.1 and 1.4.2.2 of the Tariff List (reaching a total of 30 free days); while for export containers a discount of USO 65 on the price indicated in sections 1.4.2.1 and 1.4.2.2 of the Tariff List (reaching a total of 30 free days); while for export containers a discount of USO 65 on the price indicated in sections 1.4.2.1 and 1.4.2.2 of the Tariff List (reaching a total of 30 free days); while for export containers a discount of USO 65 on the price indicated in sections 1.4.2.1 and 1.4.2.2 of the Tariff List (reaching a total of 30 free days); while for export containers a discount of USO 65 on the price indicated in sections 1.4.2.1 and 1.4.2.2 of the Tariff List (reaching a total of 30 free days); while for export containers a discount of USO 65 on the price indicated in sections 1.4.2.1 and 1.4.2.2 of the Tariff List (reaching a total of 30 free days); while for export containers a discount of 1.4.2.2 of the Tariff List (reaching a total of 30 free days); while for export

- Effective: The offer is valid from July 01, 2025 to December 31, 2025.

- Requirements: The volumes of the promotions will be accounted for during the period of validity of the offer, that is, between July 01, 2025 to December 31, 2025. For these offers there is no prior agreement or commitment of movements by the user vis-à-vis APM Terminals Callao.

### (Group 2) PROMOTION FOR REFRIGERATED CARGO CONTAINERS IN APM TERMINALS CALLAO

APM Terminals Calao (APMTC) has arranged offers for containers full with refrigerated cargo to be served by APM Terminals Calao (APMTC), either as a Port Terminal or through our Temporary Depot (code 3014), and that meet the considerations indicated lines below, depending on the type of operation 2.1 For Export containers using APMTC as Temporary Depot

- Target group: Exporters (forwarders or its representatives -customs brokers, logistics operators, etc- and other users) of containers with refrigerated cargo (reefer container)

rary offer: Refrigerated cargo exporters who insure a minimum annual volume of 1,000 reefer containers through APMTC's Temporary Depot will be exempt of the late-delivery container shipping document surcharge (documentary LAR)

- Effective: Users may apply or request this offer from May 02, 2024 until December 31, 2024, from January 3, 2025, to March 31, 2025; and from April 22, 2025, to December 31, 2025

a volumes of the offer will be accounted for annually, that is to say, in the 12 months following the date of the agreement (between APMIC and the user) to start counting the volumes (for example and as a hypothetical case; if the user's request for offer is accepted and it is agreed that the posting on June 1, 2024, then the accounting will be made until May 31, 2025). For this pumpose, us commercial area of APMIC the projections of neefer containers that will be exported in the annual period indicated above, in order to determine the availability of space in the terminal.

APMITC will carry a review every three ((3) months of the volume of containers mobilized by the user who requested the temporary offer will be cancelled, if at least 20% of the minimum annual volume requised is not neached in the first review, the temporary offer will be cancelled, if at least 20% of the minimum annual volume requised is not another the required annual volume requised is not another the required annual volume required in ordinary annual volume requir

This offer is subject to availability of space determined by APMTC.

# 2.2 For import containers that use APMTC as temporary depot (with a maximum of 8 days of stay)

- Target group: Importers of dry cargo containers (consignees or its representatives -customs brokers, logistics operators, etc.) of containers with refrigerated cargo (reefer containers) that who address cargo to APMTC's Temporary Depot.

responsible to the properties who guarantees a minimum annual movement of 200 containers (subsection 1.4.3.2), day 8 of operational area use (subsection 1.4.3.3), day 8 of reefer power supply subsection 1.5.2.4).

- Effective: The offer may be requested from August 02, 2023 to December 31, 2024; from January 3, 2025, to March 31, 2025; and from April 22, 2025, to December 31, 2025.

The minimum volumes will be counted annually, that is to say, in the 12 months following the date of the agreement (between APMTC and the user) to start counting the volumes (for example and as a hypothetical case; if the user's request for offer is accepted and it is agreed that the posting on September 1, 2023, then the accounting will be made until August 30, 2024). For this purpose, users must submit to the commercial area of APMTC the projections of containers with import refrigerated cargo that will enter DT 3014 in the annual period indicated above, in order to determine the availability of space in the terminal.

APMITC will carry a review every three ((3) months of the imported volume. If it is observed that in the first quarter at least 20% of the committed annual volume has not been met, the Offer will be cancelled; and if the review of the third quarter finds that at least 70% of the required annual volume has not been met, the Offer will be cancelled. This consideration applies to offers whose movement accounting starts on or after January 3, 2024.

For offers already agreed before January 1, 2024, the penalties described in version 12.5 or earlier of the Tariff List apply.

The acceptance of an request for access to the offer will be subject to an evaluation of the projections and/or that the user has mobilized by the Port of Calao, in the year preceding the agreement to start accounting, at least 80% of the volume committed.

In the event that the offer has been canceled to a user and the user requests access to a new offer, the rew access will be granted fire following condition is met that during the period of the cancelled offer (due to failure to comply with the volume commitment between the accounting start date and the cancelation date) it is verified that this user has handled directed by OT 3014 equal to or more than 80% of the total containers that have been directed through a temporary deposit (2014 or extraport) and unloaded loaded by North Multi-purpose Terminal.

- This offer is subject to availability of space determined by APMTC.

oort containers that use APMTC as temporary depot (with a maximum of 10 days of stay)
[arget group: Importers or associations of importers of refrigerated cargo containers (reefer containers) that who address cargo to APMTC's Temporary Depot.

- Temporary office: Importers who quarantee an annual movement of 1,000 (40-bets) refer containers will be granted a price of USD 900 for the act of services composed of the integrated temporary depot service for imported refer containers (subsection 1,4.3.2), the 8 hb 100h days of operational area use (subsection 1,4.3.3), the 8 hb 100h days of operational price of USD 900 for the act of services and price of USD 900 for the act of services and price of USD 900 for the act of services and price of USD 900 for proteins and price of USD 900 for proteins and price of USD 900 for proteins are will be changed.

- Effective: The offer may be requested from November 5, 2024, to January 31, 2025, and from April 22, 2025, to December 31, 2025.

nes will be counted annually, that is to say, in the 12 months following the date of the agreement (between APMTC and the user) to start counting the volumes (for example and as a hypothetical case; if the user's request for offer is accepted and it is agreed that the posting on No encial area of APMTC the projections of containers with import ne

APMITC will carry a review every three (03) months of the imported volume. If it is observed that in the first quarter at least 20% of the committed annual volume has not been met, the Offer will be cancelled; and if the review of the third quarter finds that at least 70% of the required annual volume has not been met, the Offer will be cancelled; and if the review of the third quarter finds that at least 70% of the required annual volume has not been met, the Offer will be cancelled; and if the review of the third quarter finds that at least 70% of the required annual volume has not been met, the Offer will be cancelled; and if the review of the third quarter finds that at least 70% of the required annual volume has not been met, the Offer will be cancelled; and if the review of the third quarter finds that at least 70% of the required annual volume has not been met, the Offer will be cancelled; and if the review of the third quarter finds that at least 70% of the committed annual volume has not been met, the Offer will be cancelled; and if the review of the third quarter finds that at least 70% of the committed annual volume has not been met, the Offer will be cancelled; and if the review of the third quarter finds that at least 70% of the committed annual volume has not been met, the Offer will be cancelled; and if the review of the third quarter finds that at least 70% of the committed annual volume has not been met, the Offer will be cancelled; and if the review of the third quarter finds that at least 70% of the committed annual volume has not been met, the Offer will be cancelled; and if the review of the third quarter finds that at least 70% of the committed annual volume has not been met, the Offer will be cancelled; and if the review of the third quarter finds that at least 70% of the committed annual volume has not been met, the Offer will be cancelled.

The acceptance of an request for access to the offer will be subject to an evaluation of the projections and/or that the user has mobifized by the Port of Calao, in the year preceding the agreement to start accounting, at least 85% of the volume con

In the event that the offer has been cancelled to a user and the user requests access to a new offer, the new access will be granted if the following condition is met: that during the period of the cancelled offer (due to failure to comply with the volume commitment between of the total containers that have been directed through a temporary deposit (3014 or extraport) and unloaded/loaded by North Multi-purpose Terminal. een the accounting start date and the cancellation date) it is verified that this user has handled/directed by DT 3014 equal to or more than 80%

ect to availability of space determined by APMTC.

# ers that use APMTC as temporary depot (with a maximum of 11 days of stay)

oms brokers, logistics operators, etc.) of containers with refrigerated cargo (reefer containers) that who address cargo to APMTC's Temporary Depot. - Target group: Importers (consignees or its rep

- Temporary offer: Importers who guarantee a minimum annual movement of 150 containers (40 feet) of refrigerated cargo will be granted a discount of USD 653.6 on the current prices, at the time of invoicing, for the package of services consisting of the integrated temporary storage service for import reefer containers (subsection 1.4.3.2), days 8 to 11 of operational area use (subsection 1.4.3.3), days 8 to 11 of reefer power supply (subsection 1.5.2.3) and days 8 to 11 of inspection and monitoring (subsection 1.5.2.4).

- Effective: The offer may be requested from May 02, 2024 to December 31, 2024, from January 3, 2025, to March 31, 2025, and from April 22, 2025, to December 31, 2025.

The minimum volumes will be accounted for annually; that is to say, in the 12 months following the date of the agreement (between APIATC and the user) to start counting the volumes (for example and as a hypothetical case; if the user's request for offer is accepted and it is agreed that the posting on June 1, 2024, then the accounting will be made until May 31, 2025). For this purpose, users must submit to the commercial area of APIATC the projections of combiners with import refrigerated cargo that will enter DT 3014 in the annual period indicated above, in order to determine the availability of space in the terminal.

APMITC will carry out a review every three (03) months of the imported volume. If it is observed that in the first quarter at least 45% of the committed annual volume has not been met, the offer will be canceled, and if the review of the third quarter finds that at least 70% of the required armial volume has not been met, the offer will be canceled, and if the review of the third quarter finds that at least 70% of the required armial volume has not been met, the offer will be canceled, and if the review of the third quarter finds that at least 70% of the required armial volume has not been met, the offer will be subject to canceled.

The acceptance of menapset to access the offer will be subject to an evaluation of the projections and/or that the user has mobilized by the Port of Calso, in the year preceding the agreement to start accounting, at least 80% of the volume committed.

In the event that the offer has been cancelled to a user and the user requests access to a new offer, the new access will be granted if the tollowing condition is met: that during the period of the cancelled offer (due to failure to comply with the volume commitm of the total containers that have been directed through a temporary deposit (3014 or extraport) and unloaded/loaded by North Multi-purpose Terminal.

This offer is subject to availability of space determined by APMTC.

# 2.5 For import containers using APMTC as temporary depot (only for new users and no penalty)

- Target group: Importers of dry cargo containers (consignees or its representatives –customs brokers, logistics operators, etc.) of containers with refrigerated cargo (reefer containers) that who address cargo to APMTC's Temporary Depot. Applies for new users.

rices consisting of the integrated temporary storage service for import reefer containers (subsection 1.4.3.2), day 8 of operational area use (subsection 1.4.3.3), day 8 of reefer p - Temporary offer: Importers who register a quarterly movement of 60 containers (40 feet) of refrigerated cargo will be granted a discount of USD 38.5 on the current prices, at the time of invoicing, for the package of se and day 8 of inspection and monitoring (subsection 1.5.2.4).

- Effective: The offer may be requested from August 02, 2023 to Dece nber 31, 2024; from January 3, 2025, to March 31, 2025; and from April 22, 2025, to December 31, 2025 - Requirements:

num volumes will be counted in the 03 months following the date of the agreement (between APMTC and the user) to start counting the volumes (for example and as a hypothetical case; if the user's request for offer is accepted and it is agreed that the posting on February 1, 2025, then the accounting will be made until April 30, 2025).

This offer is not subject to an agreement or mandatory commitment of movements by the user against APM Terminals Callao, which means that in case of non-compliance there will be no penalty. In the event that within the period of 3 months it is observed that the user does not reach the volume of the offer, then APMTC will cancel the offer to the user

### (Group 3) OFFER FOR PROMOTION FOR CONSOLIDATION/DESCONSOLIDATION OF CONTAINERS AND FOR USERS WHO HAVE ACCESSED THE PROMOTIONS OF GROUP 1 OF DT 3014

APM Terminals Calao (APMTC) has arranged offers for all those users who request a package of services and have requested and accessed the offers provided in Group 1, but referred specifically to the offers related to the user of the Temporary Depot (code 3014); that is, it does not apply to the offers of Group 1 that refer to the user of APMTC as a port terminal. Offers refer to discounts (mplying lower prices) on the aggregate price that the user would pay for the package of services consisting of any of the services specified in the Group 1 offer puls the consolidation/deconsolidation service (in any of the forms specified in subsections 1.5.3.4 to 1.5.3.3 and 1.5.3.13 of the Tariff List). The details of the offers for the package requested by the user are as detailed below.

ments or have accessed offers (set out in group 1) regarding the use of APMTC DT 3014

1) Up to 99 operations: Discount of USD 30 off the global price that would be paid with any of the modalities of the Group 1 or Group 2 offers plus the consolidation service of subsections 15.3.4 to 15.3.3 and 15.3.13 of the Tariff List, For example, if a user has accorded to numeral if of offer 1.1 or Group 1 offers (which grants a discount of USD 40 on the prices of subsections 14.1.1 or 14.1.2 and 20 days feed of strange). Dent he would have to assume a fee of USD 2755 per 278-be container and if a section of the substrainer is the subsection 15.3.4 of the Tariff List, for a board of the training of the substrainer is the substra

ii) Between 100 to 299 operations: Discount of USD 90 off the global price that would be paid with any of the modalities of the Group 1 or Group 2 offers just the consolidation deconsolidation service of subsections 1.5.3.4 to 15.3.9 and 15.3.13 of the Tariff List, For example, if a user has acceded to numeral if offer 1.1 of Group 1 offers (which grains a discount of USD 490 on the prices of subsections 1.4.1.1 or 1.4.1.2 and 20 days free of storage), then he would have to assume a fee of USD 276.6 per 20-foot container and if he additionally request deconsolidation or deconsolidation or deconsolidation or deconsolidation or perations the user will have to assume a fee of USD 276.6 per 20-foot container and if he additionally request deconsolidation or deconsoli 1.4.12 and 20 days free of storage), then he would have to assume a fee of USD 278.6 per 20-toot container and if he additionally request deconsolidation with a forklift of one of their containers, he would have to pay an additine of USD 404.6 per 20ft container (in the case of 40ft the fee with this promotion and example given will be USD 549.4)

iii) Between 300 to 799 operations: Discount of USD 70 off the global price that would be paid with any of the modalities of the Group 1 or Group 2 offers plus the con 1.4.1.2 and 30 days fiee of storage), then he would have to assume a fise of USD 268.5 per 20-bod container and if he additionally request deconsolidation with a final payment of USD 374.6 per 2010 container (in the case of 40th the elsew this trapmortion and example given will be USD 519.4) olidation service of subsections 1.5.3.4 to 1.5.3.9 and 1.5.3.13 of the Tariff List. For example, if a user has acceded to numeral iv of offer 1.1 of Group 1 offers (which grants a discount of USD 50 on the prices of subs tion with a forklift of one of their containers, he would have to pay an a

y 28 00 generation. Descript Section and Part Section (1987) and 1987 and 1

-Effective: The offers (i), ii) y ii) can be requested from August 2, 2023 to December 31, 2024. Offer iv) can be requested from March 22, 2024 to December 31, 2024. These four promotions will subsequently be eligible for applications from January 3, 2025, to March 31, 2025, and from April 22, 2025, to December 31, 2025.

Any user in requested the service package (composed of group 1 offers plus the consolidation/deconsolidation service in any of its forms) will be granted an automatic discount of USD 30 regardless of the number of consolidation/deconsolidation operations. In case users want to access the discounts of offers it and it, they will have to apply to APMITC and commit to the minimum operations numbers established in

The number of deconsolidation consolidation operations of promotors i, ii y iv will be counted considering the accounting period established in groups 1 and 2 of the offers. For example, if a user has an agreement, in group 1 or 2 of offers, whose accounting is annual and it has started on April 1, 2023 and ends on March 30, 2024, then the number of consolidation deconsidation operations will be accounted for within the last period described. Another example, if the agreement for group 1 or 2 of offers was semiannual accounting and has a start date of September 1, 2023 and an end date of February 28, 2024, then the accounting of the number of operations will be carried out within of this last period described.

APMTC will carry out a review every three (03) months of the volume of containers moved by the user who requested temporary offers it and it; it is verified, during the review, that the user will not be able to meet the required annual volume, then APMTC may cancel the offer. In this regard, if at least 15% of the minimum annual volume required is not reached in the second revision, temporary offer it and/or it will be cancelled; if at least 40% of the minimum annual volume required is not reached in the second revision, temporary offer it and/or it will be cancelled; if at least 40% of the required annual volume required is not reached in the second revision, temporary offer it and/or it will be cancelled.

APMITO will conduct a review every three ((3) months of the volume for containers mobilized by the user requesting the temporary offer (in) will apply, in this regard, if at least 25% of the minimum annual volume required is not reached in the first version, temporary offer in vivil be cancelled or, falling this, the decount from offer ii will be applied to the operations of the required annual volume, then APMITO may cancel the promotion or indicate to the user that the previous temporary offer in vivil be cancelled or, falling this, the decount from offer ii will be applied to the operations of the required annual volume, then APMITO may cancel the promotion or indicate to the user that the previous temporary offer in vivil be cancelled or, falling this, the decount from offer ii will be applied to the operations of the string upon of the required annual volume, then APMITO may cancel the promotion or indicate to the this minimum annual volume required is not reached in the first version, temporary offer in vivil be cancelled or, falling this, the decount from offer iii will be applied to the operations of the indication of the first version reached or falling this, the decount from offer iii will be applied to the operations of the string upon of the first version reached or falling this, the decount from offer iii will be applied to the operations of the string upon offer iii will be applied to the operations of the string upon of the required annual volume, the decount of offer ii will be applied to the operations of the string upon of the required annual volume, the decount of offer ii will be applied to the operations of the string upon of the required annual volume, the decount of offer ii will be applied to the operations of the string upon of the required annual volume, the decount of offer ii will be applied to the operation of the required annual volume, the decount of offer ii will be applied to the operation of the required annual volume, the decount of offer ii will be applied to the operation of This offer will be subject to space availability confirmed by APMTC

### (Group 4) OFFER FOR APPLICATION OF SERVICE PACKAGES TO CONTAINERS IN APM TERMINALS CALLAO - SECOND MODALITY

### Dear customers and users

APM Terminals Calao (APMTC) has set special offers for all those dry import containers that are served through the Port Terminal or through its Temporary Depot (IDT 3014) for which service packages are requested. The offers refer to discounts (which imply lower prices) and apoly by package how which are detailed below:

- Target group: Users who do not have agreements or have not accessed DT 3014 offers (Groups 1 or 2)
- Temporary offer it: without the need to reach a certain volume
  - Group of packets 1 DT 3014: integrated service for 20 or 40 froit import dy container (subsections 1 4.1.1 and 1 4.1.2 of the Tariff List) and container deconsolidation in any of its forms (subsections 1 5.3.5, 4, 15.3.5 and 1 5.3.5 of the Tariff List). For packages that are desired, a global discount of USD 30 for a 20-bot container or 40-bot container will be granted.
  - Group of packets 2 DT 3014: integrated service for 20 or 40 foot import dry container (subsections 1.4.1.1 and 1.4.1.2 of the Tariff List), container deconsolidation in any of its forms (subsections 1.5.3.4, 1.5.3.5 and 1.5.3.6 of the Tariff List) and use of area from day 11 to day 20 (subsection 1.4.1.3 of the Tariff List). For packages that are claimed, a global discount of USD 183.1 for a 20-foot container and USD 371.7 for a 40-foot container will be greated.
  - Forum of packets 3 DT 3014: integrated service for 20 or 40 foot import dry container (subsections 1.4.1.1 and 1.4.1.2 of the Tariff List), and partial consolidation of container with dry cargo (subsection 1.5.3.13 of the Tariff List), For packages that are claimed, a discount of USD 25 for a 20-foot container or 40-foot container or 40-foot container.
  - Group of packets 4 as port terminat: cargo portion of Standard Service for 20 or 40 foot import dry container (subsections 1.1.2.1 and 1.1.2.2 of the Tariff List). For packeges that are daimed, a global discount of USD 29.6 for a 20-toot container and USD 39.5 for a 40-toot container will be granted.
  - Group of packets 5 as port terminal: argo portion of Standard Service for 20 or 40 foot import dy container (subsections 1.1.2.1 and 1.1.2.2 of the Tariff List) and partial consolidation/deconsolidation of container with dy cargo (subsection 15.3.13 of the Tariff List). For packages that are claimed, a global discount of USD 13.6 for a 20 or 40 foot container will be granted
  - offer ii): based on demand or volume: for users who carry out more than 200 deconsolidation operations in an annual period

Group of packets 2 - DT 3014: integrated service for 20 or 40 foot import dry container (subsections 1.4.1.1 and 1.4.1.2 of the Tariff List), container deconsolidation in any of its forms (subsections 1.5.3.4, 1.5.3.5 and 1.5.3.6 of the Tariff List) and use of area from day 11 to day 20 (subsection 1.4.1.3 of the Tariff List). For this package and based on the minimum number of operations required, an global disc

Effective: The offer ii) is valid from Auoust 02, 2023 to December 31, 2024 from January 3, 2025 to March 31, 2025 and from Aoril 22, 2025 to June 30, 2025. The offer ii) may be requested from Auoust 02, 2023 to December 31, 2024 from January 3, 2025 to March 31, 2025 and from Aoril 22, 2025 to June 30, 2025.

The number of operations of the temporary offer it will be counted within the annual period agreed between APMITC and the user, the earliest date to start the accounting may be April 1, 2023 (for example and as a first hypothetical case; if the user's offer request is accepted and it is agreed that the accounting will start on 1 July 2023, then the accounting will start on 1 July 2023, then the accounting will start on 1 July 2023. ample, if the user's offer request is accepted and it is agreed that the accounting will begin on September 1, 2023, then the accounting will be carried out until August 31, 2024).

APMIC will carry out a review every three (03) months of the volume of containers moved by the user with orequested temporary offers; if it is verified, during the review, that the user will not be able to meet the required annual volume, then APMIC may cancel the offer. In this regard, if at least 15% of the minimum and 40% of the iminimum annual volume required is not reached in the second revision, temporary offer it will be cancelled.

in case of changes in the prices of the Tariff List subsections involved in these offers (1.1.2.1, 1.1.2.1.4.1.1, 1.4.1.2.1.4.1.3, 1.5.3.4, 1.5.3.5, 1.5.3.5 y 1.5.3.13), the discounts described apply to these modified prices in force at the time of involoning. See invoicing details in the Regulation of Tariffs and Commercial Policy of APM Terminals Calloo.

This offer will be subject to space availability confirmed by APMTC

### (Group 5) OFFER FOR UN 3077 CARGO CONTAINERS CLASSIFIED AS IMO 9 IN APM TERMINALS CALLAO'S TEMPORARY DEPOT

APM Terminals Callac (APMITC) has set that export containers with dangerous cargo UN 3077 (UN 3077) - such as zinc oxide, abletic acid and others - described as IMO 9 class, to be serviced through its Temporary Depot (code 3014), the price of the special service "Treatment of dangerous cargo IMO 9" will not be applied; in other words, a 100% discount will be applied to the price of subsection 1.5.1.3. This promotion will be valid from January 1, 2021 to Marria 13, 2025, and from April 11, 2025, by December 31, 2025.

# (Group 6) SPECIAL OFFER FOR FISHMEAL IN CONTAINERS

# Dear customers and users

APM Terminals Callao (APMTC) has set offers for containers with fishmeal. These offers are governed under the following terms:

# 6.1 First promotion - as port terminal; exemption from the price of the special service "high cube containers"

Any user who loading fishmeal in containers will not be charged the grice of the special service "high cube containers" of subsection 1.5.1.8. This offer will be valid from July 1.2021 to March 31, 2025, and from April 11, 2025 to December 31, 2025.

# 6.2 Second promotion - as port terminal or temporary depot: discount on standard service tariff and on special service price of special treatment of Hazardous cargo

- Target group: Exporters (forwarders or its representatives -customs brokers, logistics operators, etc- and other users) of of 40-foot containers with fishmeal
- Emproarry offer. Fishmeal exporters shall be granted discounts on the standard cargo-portion service tariffs actually changed (subsection 1.1.2.4 of the Tariff List) or on the integrated service for temporary storage of 40-foot export containers (subsection 1.4.2.2 of the Tariff List), on the price of the special service for special realment of hazardous cargo IMO9 (subsection 1.5.1.2 or 1.5.1.3 of the Tariff List); The details of the discounts are as follows:

(ii) As port terminal - For any user who loading fishmeal: A discount of USD 96.7 will be granted on the general promotion teriff of standard carpo-portion service applicable to full 40-foot containers (subsection 1.1.22 of the Tariff List) that is in force at the time to request standard service. Likewise, the payment of the special service for treat 15.1.2 or 15.1.3 of the Tariff List) shall be exempted. Includes exemption from payment of the price for the special injectuce service (subsection 15.1.8).

As Port Terminal and Temporary Depot (DT) 3014 - Aysu service in the process of t

iii) As Port Terminal and Temporary Depot (IDT) 3014. Any user who load featment on a shipping line that, during the first half of 2005, has loaded 5,000 TEUs will be granted a discount of USD 107.1 on the standard service promotion lastfill land portion applicable to full 40-box containers (subsection 1.1.2 of the Tarff List) and a discount of USD 110.7 on the price of the in temporary depot service for 40-box depot containers (subsection 1.2.2 of the Tarff List) and a discount of USD 110.7 on the price of the in temporary depot service for 40-box depot containers (subsection 1.2.2 of the Tarff List) and a discount of USD 110.7 on the price of the in temporary depot service for 40-box depot depot services for special services for spe

ii) As Port Terminal and Temporary Depot ([17]) 3014: Later, if the shipping line, which is found in items ii) or iii, reaches 10,000 TEU (counted from January 1, 2025) at any time in 2025, the user (exponder) will be granted a discount of USD 117.1 on the general promotional listiff for standard service land-portion applicable to full 40-bot containers (subsection 1.1.2.4 of the Tariff List) and a discount of USD 112.7 on the integrated temporary depot service price to 47-bot depot export containers (subsection 1.1.2.2 of the Tariff List) and a discount of USD 112.7 on the integrated temporary depot service price to 47-bot depot export or the special service for special seatment of disrepances cargo (subsections 1.5.1.2 or 15.1.3 of the Tariff List) will be waited. Export the special service for the special service for special seatment of the special special service for the specia

v) As Port Terminal and Temporary Depot (ITI) 3014: Later, if the shipping line, which is found in item iv), reaches 15,000 TEU (counted from January 1, 2025) at any time in 2025, the user (exponer) will be granted a discount of USD 125.1 on the general promotional lateff for standard service land-portion applicable to full 49-foot containers (subsection 1.1.2.4 of the Tariff List) and a discount of USD 103.7 on the integrated temporary depot pervice 3014; also, the payment of the special service for special treatment of dangerous cargo (subsections 1.5.1.2 or 1.5.1.3 of the Tariff List) will be waived; this payment of the special service for the special service for 15 special point counts will apply to contain some of the fine date on which 10,000 TEUs are readed until December 31, 2025.

titive. The offer i) is valid from August 15, 2025 to December 31, 2025. The offers i), ii), iv) and v) will take effect on August 22, 2025 (ending on December 31, 2025), but the discounts from promotion iii) will apply to containers that have been moved between July 1, 2025, and December 31, 2025

# (Group 7) OFFER FOR MINERAL LOADED IN CONTAINERS

APM Terminals Calao (APMTC) has set offers for loading containers with mineral concentrates (bulk) and/or minerals in big bags and minerals with certain added value (such as ingots, bars, jumbos, etc.) for loading. These offers are governed by the following terms:

- Target group: Exporters (forwarders or its representatives -customs brokers, logistics operators, etc. and other users) of of 20-foot containers with mineral concentrates, minerals in big bags and minerals with certain added value (such as ingots, bars, jumbos, etc.)

- Temporary offer: Exporters of mineral concentrates, minerals in big bags and minerals with a certain added value will be granted discounts on the general promotional tariff for standard service lend-portion actually changed (subsections 1.1.2.3 and 1.1.2.4 of the Tariff List) or on the integrated service prices for temporary depot of 20- and 40-foot dry export containers (subsections 1.4.2.1 and 1.4.2.2 of the Tariff List), on the price of the special service for the special serv

(i) As port terminal - Any user who bading mineral concentrates, minerals in the bags and minerals with a certain added value: A discount of USD 50.2 will be granted on the general promotional tariff of standard service land-portion applicable to 1.81 20-foot containers (subsection 1.1.2.3 of the Tariff List) and a discount of USD 54.6 on the general promotional tariff of standard service cargo-portion applicable to 4-floot dry full containers (subsection 1.1.2.4 of the Tariff List) in effect at the time of requesting the standard service. Likewise, the payment of the special service for special terestener of dangerous cargo will be exempt (subsection 1.5.1.3 of the Tariff List).

(ii) As Temporary Depot (107) 3014 - Any user who badding mineral concentrates, mineral some interesting the payment of the special service of containers and USD 55.7 for 40-foot containers will be applied to the integrated service prices for temporary storage of dry export containers (subsections 1.4.2.1 and 1.4.2.2 of the Tariff List); a total of 14 free days of user of the questing area will also be granted, it includes exemption from payment of the special service or special terminated of the special service or special terminated of dangerous cargo will be exemption to the integrated service prices for temporary storage of dry export containers (subsections 1.4.2.1 and 1.4.2.2 of the Tariff List); a total of 14 free days of user of the operating area will also be granted, it includes exemption from payment of the special service or containers will be applied to the integrated service prices for temporary storage of dry export containers (subsections 1.4.2.1 and 1.4.2.2 of the Tariff List); a total of 14 free days of user of the part of the special service prices for temporary storage of dry export containers (subsections 1.4.2.1 and 1.4.2.2 of the Tariff List); a total of 14 free days of user of the part of the payment of the special service prices for the part of the part of the payment of the payment of the payment of the paymen

ii) As Port Terminal and Temporary (Dept) (III) 3014: Any users in this target group who load on a shipping line that reaches, at any time during the year 2025, a movement of 6,000 TEUs, will be granted a series of discounts and benefits. These discounts and benefits will apply to containers shipped from the date the 6,000 TEU threshold is reached until December 31, 2025. The

Containers entering as a Port Terminal will be granted a discount of USD 68.2 on the general promotional tariff of standard service land-original policy following for the containers of export (subsection 1.1.2.3 of the Tariff List) and a discount of USD 75.1 the general promotional tariff for standard service cargo-cortion applicable to 40-foot dry full containers for export (subsection 1.1.2.3 of the Tariff List) and a discount of USD 75.1 the general promotional tariff for standard service cargo-cortion applicable to 40-foot dry full containers for export (subsection 1.1.2.3 of the Tariff List) and a discount of USD 75.1 the general promotional tariff for standard service cargo-cortion applicable to 40-foot dry full containers for export (subsection 1.1.2.3 of the Tariff List) and a discount of USD 75.1 the general promotional tariff for standard service cargo-cortion applicable to 40-foot dry full containers for export (subsection 1.1.2.3 of the Tariff List) and a discount of USD 75.1 the general promotional tariff for standard service cargo-cortion applicable to 40-foot dry full containers for export (subsection 1.1.2.3 of the Tariff List) and a discount of USD 75.1 the general promotional tariff for standard service cargo-cortion applicable to 40-foot dry full containers for export (subsection 1.1.2.3 of the Tariff List) and a discount of USD 75.1 the general promotional tariff for standard service cargo-cortion applicable to 40-foot dry full containers for export (subsection 1.1.2.3 of the Tariff List) and a discount of USD 75.1 the general promotional tariff for standard service cargo-cortion applicable to 40-foot dry full containers for export (subsection 1.1.2.3 of the Tariff List) and a discount of USD 75.1 the general promotional tariff for standard service cargo-cortion applicable to 40-foot dry full containers for export (subsection 1.1.2.3 of the Tariff List) and a discount of USD 75.1 the general promotion applicable to 40-foot dry full containers for export (subsection 1.1.2.3 of the Tariff List

(subsection 1.1.2.4 of the Tariff List).

. Containers entering 07 3014 will be granted a discount of USD 72.1 on the integrated service price for temporary depot of 20-foot dry full containers for export (subsection 1.4.2.1 of the Tariff List), and a discount of USD 65.7 on the price of the integrated service for temporary depot of 40-foot full containers (subsection 1.4.2.2 of the Tariff List). Additionally, containers destined for DT 3014 will receive a total of 14 free days of operational area usage. Both containers entering the port terminal or being directed to DT 3014 are exempt from paying the special service for special handling of dangerous cargo (IMO 9).

ii) As Port Terminal and Temporary Depot (DT) 3014: Any users in this target group who load on a shipping line that, during the first half of 2025, has reached a throughput of 6,000 TEUs will be granted the same benefits outlined in item iii) of the offers. These discounts and benefits will apply to containers shipped between July 1, 2025, and December 31, 2025. The discounts are described below.

Containers entering as a Port Terminal will be granted a discount of USD 69.2 on the general promotional tariff of standard service cargo-portion applicable to 20-foot dry full containers for export (subsection 1.1.2.3 of the Tariff List) and a discount of USD 75.1 the general promotional tariff or standard service cargo-portion applicable to 40-foot dry full containers for export (subsection 1.1.2.4 of the Tariff List) and a discount of USD 75.1 the general promotional tariff or standard service cargo-portion applicable to 40-foot dry full containers for export (subsection 1.1.2.4 of the Tariff List). passession. In Law time is am Law, contained and secount of USD 72.1 on the integrated service price for temporary depot of 20-foot dry full containers entering DT 3014 will be granted a discount of USD 65.7 on the price of the integrated service for temporary depot of 40-foot full containers destinated for DT 3014 will receive a total of 14 free days of operational area usage.

Both containers entering the port terminal or being directed to DT 3014 are exempt from paying the special service for special handling of dangerous cargo (IMO 9).

v) As Port Terminal and Temporary Depot (IDT) 3014: Lister, file elipsion line, which is found in items ii) or iv), resorbes 12,000 TEUs (counted from January 1, 2025), at any point during 2025, the user (exporter) will be granted benefits exceeding those specified in items iii) and iv). These discounts and benefits will apply to containers loaded from the date the 12,000 TEUs (resorbed vision). Containers entering as a Port Terminal will be granted a discount of USD 76.2 on the general promotional tariff of standard service land-portion applicable to 20-tool dry full containers for export (subsection 1.1.2.3 of the Tariff List) and a discount of USD 85.1 the general promotional tariff for standard service cargo-portion applicables to 20-tool dry full containers for export (subsection 1.1.2.4 of the Tariff List) and a discount of USD 85.1 the general promotional tariff for standard service cargo-portion applicables to 20-tool dry full containers for export (subsection 1.1.2.4 of the Tariff List) and a discount of USD 85.1 the general promotional tariff for standard service cargo-portion applicable to 20-tool dry full containers for export (subsection 1.1.2.3 of the Tariff List) and a discount of USD 85.1 the general promotional tariff for standard service cargo-portion applicable to 20-tool dry full containers for export (subsection 1.1.2.4 of the Tariff List) and a discount of USD 85.1 the general promotional tariff for standard service cargo-portion applicable to 20-tool dry full containers for export (subsection 1.1.2.4 of the Tariff List) and a discount of USD 85.1 the general promotional tariff for standard service cargo-portion applicable to 20-tool dry full containers for export (subsection 1.1.2.4 of the Tariff List) and a discount of USD 85.1 the general promotional tariff for standard service cargo-portion applicable to 20-tool dry full containers for export (subsection 1.1.2.4 of the Tariff List) and a discount of USD 85.1 the general promotional tariff for standard service cargo-portion applicable to 20-tool dry full containers for export (subsection 1.1.2.4 of the Tariff List) and a discount of USD 85.1 the general promotion applicable to 20-tool dry full containers for export (subsection 1.1.2.4 of the Tariff List) and a discount of USD 85.1 the general promotion applicable to 20-tool dry full containers for export (subsection 1.1.2.4 of the Tariff List) and a discount of USD 85.1 the gen

Containers entering DT 3014 will be granted a discount of USD 80.1 nn the integrated service for temporary depot of 40-foot fly full containers for export (subsection 1.4.2.1 of the Tariff List) and a discount of USD 83.7 on the price of the integrated service for temporary depot of 40-foot full containers (subsection 1.4.2.2 of the Tariff List). Additional residence for DT 3014 will receive a total of 14 free days of operational area usage.

Both containers entering the port terminal or being directed to DT 3014 are exempt from paying the special service for special handling of dangerous cargo (IMO 9).

(v) As Port Terminal and Temporary Depot (IDT) 3014; Liste, if the abligning line, which is found in item v), reaches 15,000 TEU (rounded from January 1, 2025), at any point during 2025, the user (exporter) will be granted benefits exceeding those specified in items ii), iv) and v). These discounts and benefits we indicate the second of the second of

Containers entering as a Port Terminal will be granted a discount of USD 86.2 on the general promotional tarriff of standard service land-portion applicable to 20-foot dry full containers for export (subsection 1.1.2.4 of the Tarriff List) and a discount of USD 95.1 the general promotional tarriff for standard service cargo-portion applicable to 40-foot dry full containers for export (subsection 1.1.2.4 of the Tarriff List) and a discount of USD 95.1 the general promotional tarriff for standard service cargo-portion applicable to 40-foot dry full containers for export (subsection 1.1.2.4 of the Tarriff List) and a discount of USD 95.1 the general promotional tarriff for standard service cargo-portion applicable to 40-foot dry full containers for export (subsection 1.1.2.4 of the Tarriff List) and a discount of USD 95.1 the general promotional tarriff for standard service cargo-portion applicable to 40-foot dry full containers for export (subsection 1.1.2.4 of the Tarriff List) and a discount of USD 95.1 the general promotional tarriff for standard service cargo-portion applicable to 40-foot dry full containers for export (subsection 1.1.2.4 of the Tarriff List) and a discount of USD 95.1 the general promotional tarriff for standard service cargo-portion applicable to 40-foot dry full containers for export (subsection 1.1.2.4 of the Tarriff List) and a discount of USD 95.1 the general promotional tarriff for standard service cargo-portion applicable to 20-foot dry full containers for export (subsection 1.1.2.4 of the Tarriff List) and a discount of USD 95.1 the general promotional tarriff for standard service cargo-portion applicable to 20-foot dry full containers for export (subsection 1.1.2.4 of the Tarriff List) and a discount of USD 95.1 the general promotional tarriff for standard service cargo-portion applicable to 20-foot dry full containers for export (subsection 1.1.2.4 of the Tarriff List) and a discount of USD 95.1 the general promotion applicable to 20-foot dry full containers for export (subsection

s entering DT 3014 will be granted a discount of USD 90.1 cm the integrated service price for temporary depot of 40-foot full containers (subsection 1.4.2 of the Tariff List), and a discount of USD 97.7 on the price of the integrated service for temporary depot of 40-foot full containers (subsection 1.4.2 of the Tariff List), destined for DT 3014 will receive a total of 14 fee days of operational area usage.

Both containers entering the port terminal or being directed to DT 3014 are exempt from paying the special service for special handling of dangerous cargo (IMO 9).

- This offer will be subject to space availability confirmed by APMTC

### (Group 8) OFFER FOR TRANSSHIPMENT OF CONTAINERS

### Dear customers and users,

APM Terminals Calao (APMTC) has set offers for transshipment containers that are loaded/unloaded by the North Multi-purpose Terminal. These offers are governed by the following terms:

### 8.1 First offer - 800 containers: discount on standard service, 14 days free of use of operational area (possible extension) and exemption from change of status

- Target group : Users (shipping lines or its representatives) that move transhipment containers through the North Multi-purpose Terminal
- Temporary offer: Users who move transshipment containers equal to or greater than 800 weekly average containers, over a period of one year, will be granted a discount on the standard container transshipment service tariff and an extension of free days, as follows:

(i) > 800 serage containers per use-selt: They will be disposate a standard service leaff of USD 90 for 20-bot transahipment containers (subsection 1.1.2 of the Taiff Lis), it includes for both sizes of containers (20 and 40 feet) the garring of 14 fee days (which register leaff of USD 90 for 20-bot transahipment containers (subsection 1.1.2 of the Taiff Lis), it includes for both sizes of containers (20 and 40 feet) the garring of 14 fee days (which register leaff of USD 90 for 20-bot transahipment containers (subsection 1.1.2 of the Taiff Lis), it includes for both sizes of containers (20 and 40 feet) the garring of 14 feet days (which register) and the garring of

(ii) It includes for both o tainer sizas (20 and 40 feet) the granting of 14 free days (which represents 12 days off in addition to the 48 hours off currently granted). APMTC may extend, but subject to space availability, to 21 days free of use of the operational area.

- Effective: The offers i) and i) is valid from August 15, 2025 to December 31, 2025. The offers ii), iv), v) and vi) will take effect on August 22, 2025 (ending on December 31, 2025), but the discounts from offer iv) will apply to containers that have been moved between July 1, 2025, and December 31, 2025.

(iii) Likewise, users who comply with the weekly average movement of 800 containers, during an annual period, will be exempted from the payment of the container status change surcharge

ective: The offer may be requested from November 15, 2023 to January 31, 2024, and from January 3, 2025, to March 31, 2025.

The average weekly volume shall be accounted and calculated within the annual period agreed between APMITC and the user (for example and as a hypothetical case, if the user's request for offer is accepted and it is agreed that the accounting will start on December 1 2023, then the accounting will be carried out until November 30, 2024. All types of Inanshipment (full cycle or interminal) shall be accounted. Accounting is per container, for example, during this ammunal period the user outdomners are 40°s letter-from a full containers, which add up to a weekly average of 1500 or dariners (which is said to the accounting will be a created out until November 30, 2024.) All types of Inanshipment (full cycle or interminal to the period (which is 10°s) as weekly average of 1500 or dariners (which is said to the accounting will be a result of the accounting and the accounting will be a result of the accounting will be a created or the accounting will be a result of the accounting will be a result of the accounting and account the accounting will be a result of the accounting will be a result

APMTC will carry out a review every three (03) months of the weekly average of the transshipment containers mobilized by the user who requested the offer. If it is verified, during the review, that the user has not compiled with the required weekly average, then APMTC may cancel the offer

- This offer will be subject to space availability confirmed by APMTC

### 8.2 Second offer - 1.300 containers: two types of discounts on standard service. 14 days free of use of operational area, free pool and exemption from change of status

- Target group : Users (shipping lines or its representatives) that move transhipment containers through the North Multi-purpose Termina

- Temporary offer: Uses who move transshipment containers equal to or greater than 1,300 weekly average containers, over a period of 2 years and 3 months (09 quarters or 27 months in total), will be granted similar discounts or benefits as offer 8.1 but an additional discount is added to the standard service terriff that will be reflected in discount amounts and other additional benefits. The details of the discounts and benefits of the associated as a Solivance.

(i) Change of an initial offer tartif of USO 90 for standard 20-boot transhipment container service (Section 1.1.2.7 of the Tartif List) and USO 120 for standard 40-boot transhipment container service (Section 1.2.2.7 of the Tartif List).

ii) Additional discount of USD 35 per container on the offer tariffs described in item ii), but which may be given for a maximum of 9,143 containers in each quarter during during the 99-quarter (or 27-month) period of the offer. This additional discount, wich will applied during the period of the offer, will be for a one-time only, that is, it will not apply in case of nenewal of this offer.

iii) Granting of 14 free days of use of the operational area (which represents 12 additional free days to the 48 free hours currently granted) for both container sizes (20 and 40 feet).

iv) Granting of a free pool of 500 TEU's (use of operational area of 500 TEU's of transhipment without time limit). This free pool is granted to those containers that have exceeded the 14 days of use of the op

v) Exemption from payment of container status change surcharge.

vi) Additional discount of USD 25 per container on the promotional tariffs described in item (i), but which may be given for a maximum of 3,976 containers and only in the first quarter of the offer. This discount does not apply for the remaining eight (08) quarters and will not apply in the event of renewal of the offer

-Effective: The offer may be requested from January 03, 2024 to March 31, 2024, and from October 1 to October 30, 2024. For users who have requested the offer before March 31, 2024, the free pool of 500 TEU (item in of this offer) will be offered from October 1, 2024.

The average weekly volume will be counted and calculated within the period of 2 years and 3 months agreed by APMTC and the user, which may be the earliest date to start counting on October 1, 2022 (for example and as a hypothetical case; if he user's request for offer is accepted and it is agreed that the accounting will start on October 1, 2022, then he accounting will be carried out until December 2025. As a second hypothetical example, if the user's offer request is accepted and it is agreed to start accounting on February 01, 2024, then accounting will be done until April 30, 2026). All types of transhipment [full cycle or inter-terminal] shall be accounted. Accounting is per container; to recample, during the accounting period the user could move a weekly average of 1,200 full cycle containers and 1 inter-terminal containers, which add up to a weekly average of 1,300 containers (which results in a weekly average of 2,500 movements; 2,400 full cycle movements and 100 inter-terminal movements). The average weekly count is calculated by taking into account the total number of transhipment containers moved during the count period and divided by the number of weeks in that period (which is 117 weeks). For this purpose, users should submit to the APMITC commercial area the projections of transhipment containers that they will be mobilize during the period indicated above, in order to determine the availability of terminal space.

APMITC will carry out a review every three (03) months of the weekly average of the transhipment containers mobilized by the user requesting the offer, if it is verified, during the review, that the user has not met the required weekly average, then APMITC may cancel the offer.

APMITC will carry out a review every three ((3) months of the weekly average of the transshipment containers mobilized by the user requesting the offer, if its verified, during the review, that the user has been complying with, at least 90% of the nequired weekly average, then APMITC will apply for that quarterly period and with respect to the item it discount, the additional discount of USO 35 (on the initial promotional tariffs for standards service of containers of item) over a maximum of 9,143 transshipment containers. This quarterly discount, which will apply to a maximum of 9,143 containers, will be granted in the form of a credit role or another modelity agreed upon by the shipping line and APMITC.

onal discount of USD 25 per container set out in item vi) will be granted as long as the user has complied, during the first quarter, with the committed weekly average (that is, with 1,300 weekly average containers). This discount will only be granted for the first quarter of the offer.

In case of containers that exceed the 14 days free of use of the operating area and the free pool of 500 TEU, the user will have to assume the payment of the tairff applicable from the 15th day orwards (subsections 1.2.6.4 and 1.2.7.4 of the Tairff List), However, if the user mobilizes more than 1,300 containers per week on average, APMITC may extend, subject to space availability, the days of use of the operational area for those containers that remain more than 14 days and cannot access the free pool. The extension days would be up to seven days at most. If APMITC grants this benefit to a user, it will be applied to other users under similar conditions.

will be subject to space availability confirmed by APMTC

### rd offer - 1190, 1360 and 1530 containers: two types of discounts on standard service, 14 days free of use of operational area, free pool and exemption from change of status

- Target group : Users (shipping lines or its representatives) that move transhipment containers through the North Multi-purpose Terminal

- Temporary offer: Users who move transhipment containers of 1,190, 1,360, and/or more than 1,530 average containers per week during the second quarter of 2025 (from April 1, 2025, to June 30, 2025) will be granted similar discounts or benefits as those provided in Offer 8.1, but with a higher rebate amount on the additional discount on the additional discount on the standard service tariff. The details of the discounts or benefits that apply to this promotion are as follows:

i) Charge of an initial offer tariff of USD 90 for standard 20-foot transhipment container service (Section 1.1.2.7 of the Tariff List) and USD 120 for standard 40-foot transhipment container service (Section 1.1.2.8 of the Tariff List)

ii) Additional discounts per container on the offer tariffs described in item i), but which may be given for a maximum number of containers in the promotional accounting quarter. The additional discounts are as follows

ii.1) If a weekly average of 1.190 to 1.359 containers is reached: Additional discount of USD 35 per container on the offer tariffs described in item ii), but which may be given for a maximum of 9.143 containers in the quarter in which the offer is recorded

ii.2) If a weekly average of 1,360 to 1,529 containers is reached: Additional discount of USD 50 per container on the offer tariffs described in item i), but which may be given for a maximum of 12,800 containers in the quarter in which the offer is recorded ii.3) If a weekly average of 1,530 containers is reached and/or exceeded: Additional discount of USD 70 per container on the offer tariffs described in item i), but which may be given for a maximum of 13,175 containers in the quarter in which the offer is recorded.

iii) Granting of 14 free days of use of the operational area (which represents 12 additional free days to the 48 free hours currently granted) for both container sizes (20 and 40 feet).

iv) Granting of a free pool of 500 TEU's (use of operational area of 500 TEU's (use of operational area of 500 TEU's of transhipment without time limit) during the promotional period. This free pool is granted to those containers that have exceeded the 14 days of use of the operational area

v) Exemption from payment of container status change surcharge.

- Effective: The offer may be requested from April 30, 2025 to June 1, 2025

The average weekly volume will be counted and calculated during the second quarter of 2025 (from June 30, 2025, hrough June 30, 2025). Accounting is per container, for example, during the accounting period, a user could move a weekly average of 1,430 full cycle containers and 100 inter-terminal movements). The average weekly count is calculated by taking into account the total number of transhipment containers moved during the count period and divided by the number of weeks in that period (which is 13 weeks). For this purpose, users should submit to the APMITC containers moved during the period indicated above, in order to determine the availability of terminal space. erage of 1.530 containers (which results in a weekly a nercial area the projections of transhipment co

APMTC will carry out a review every 15 days of the weekly average of the transshipment containers mobilized by the user requesting the offer, if it is verified, during the review, that the user has not met or reached the weekly average committed, then APMTC will propose the offer of the lower range (for example, if the user committed to reaching a weekly volume of 1,530 and reaches a vol. APMTC will propose the application of the offer with the range of 1,300 to 1,529 containers) or may cancel the offer.

The discount for item ii (of USD 35, USD 50 or USD 70) will be granted in the form of a credit note or other modality agreed upon by the shipping line and APMTC.

In case of containers that exceed the 14 days of free use of the operational area and the free pool of 500 TEU, the user will have to assume the payment of the tariff applicable from the 15th day onwards (subsections 12.6.4 and 12.7.4 of the Tariff List), However, if the user mobilizes more than 1,300 containers per week on average, APMTC may extend, subject to space availability, the days of use of the operational area for those containers that remain more than 14 days and cannot access the free pool. The extension days would be up to seven days at most. If APMTC grants this benefit to a user, it will be applied to other users under similar conditions.

This promotion may also be a reliable to shipping lines that have existing agreements for other offers is Group 8. For shipping lines that access and meet the volume ranges required by this offer, and in order not to compountse the volume fulfillment they have agreed to in other offers, it will be granted that the movements made during the second quarter of 2025 (between April 1, 2025, and J. apply or be accounted for in offers 8.1 or 8.2, but up to a maximum. If the user has the 8.1 offer in effect, they will be allowed to assign up to 9.00 containers per week, 2 flery have the 8.2 offer in effect, they will be allowed to assign up to 1,300 containers per week, 2 flery have the 8.2 offer in effect, they will be allowed to assign up to 1,300 containers per week, 2 flery have the 8.2 offer in effect, they will be allowed to assign up to 1,300 containers per week, 2 flery have the 8.2 offer in effect, they will be allowed to assign up to 1,300 containers per week, 2 flery have the 8.2 offer in effect, they will be allowed to assign up to 1,300 containers per week, 3 flery have the 8.2 offer in effect, they will be allowed to assign up to 1,300 containers per week, 3 flery have the 8.2 offer in effect, they will be allowed to assign up to 1,300 containers per week, 4 flery have the 8.2 offer in effect, they will be allowed to assign up to 1,300 containers per week, 4 flery have the 8.2 offer in effect, they will be allowed to assign up to 1,300 containers per week, 4 flery have the 8.2 offer in effect, they will be allowed to assign up to 1,300 containers per week, 4 flery have the 8.2 offer in effect, they will be allowed to assign up to 1,300 containers per week, 4 flery have the 8.2 offer in effect, they will be allowed to assign up to 1,300 containers per week, 4 flery have the 8.2 offer in effect, they will be allowed to assign up to 1,300 containers per week, 4 flery have the 8.2 offer in effect, they will be allowed to assign up to 1,300 containers per week, 4 flery have the 8.2 offer in effect, they will b

- This offer will be subject to space availability confirmed by APMTC

# (Group 9) OFFER FOR GASIFICATION OF FULL CONTAINERS

# Dear customers and users

APM Terminals Calao (APMTC) has set a offer for volume for those users (shipping lines) that demand the qasification service of full containers that requires the joint supply of N2 and CO2 as a port terminal. This offer is governed by the following terms

(i) ≥ 150 containers quarterly = Tariff of USD 495 which will include the following: combined gasification service of N2 and CO2, change of status (in case of occurrence) and the benefit of granting customers (exporters) of shipping lines 2 free days of use of the operating area in addition to the 72 hours granted as part of the standard shipping container service (up to a total of 5 free days of use of the operating area as a Post Terminal).

Effective: The offer with the new price of USD 495 will be available from July 1, 2024 to December 31, 2024, and from Jeruary 3, 2025, to March 31, 2025, and from April 22, 2025, to June 30, 2025. For offers that have been agreed up to 30 June 2024, the price set out in version 12.10 of the Tariff applies

The minimum volumes of promotions will be counted on a questerly basis; that is, in the 33 months following the date of the agreement (between PAPICT and the user) to dart counting the volumes for example and as a hypothetical case; if the user's request for promotion is accepted and it is agreed that the accounting on April 1, 2024, then the accounting will be carried out until June 30, 2024; For this purpose, user must admit to the commercial area of APICT (be propercious of continenent that will demand the applications service in the eximation.)

If, after the three ((3) months of accounting, it is verified that the minimum volumes required for the application of these offers have not been reached, APMTC will charge the maximum tariff in subsection 12.13.1 in effect at the time of verification and the charge of the charge of status surcharge (subsection 9.3.1.1 of the Tariff List) in effect at the time of billing will apply,

- This offer will be subject to space availability confirmed by APMTC

# (Group 10) OFFER FOR SHIPPING LINES - EMPTY CONTAINERS

APM Terminals Callao (APMTC) has set offers for empty loading (exported) containers; these offers are governed by the following terms:

# 10.1 Extension of free stay days for empty containers and users with large volumes

- Target group: Users (shipping lines or their representatives) that move empty export containers through the Multipurpose North Termina

- Temporary offer: Shipping lines that move, during an annual period, a volume equal to or greater than 25,000 empty export containers will be granted the right to request, for each vessel, an extension of 24 hours of fee time that are additional to the hours of use of operating area free set out in subsection 12.5.1 of the Tariff List (which are 72 hours of free time), which means that shipping lines that volume of containers indicated above will, in principle, be granted 96 hours of free time of use of the operating area. The detail of the offer is as follows:

(i) Movement of 500 empty export containers per vessel = Empty containers on this vessel will be granted a total of 96 hours of free operating area use (which represents an extension of 24 hours free from the 72 hours free granted under subsection 1.2.5.1 of the Tariff List).

(ii) Movement of 1,000 empty export containers per vessel = Empty containers on this vessel will be granted a total of 120 hours of free operating area use (which represents an extension of 48 hours free from the 72 hours free granted under subsection 1,2,5.1 of the Tariff List).

- This offer will be subject to space availability confirmed by APMTC

The free firm of 95 or 120 hours of operational area use will have be requested by each vessel that docks. In those cases where the shipping line did not request the application of the offer to a particular vessel, then that vessel will only be granted the free time set out in subsection 1.2.5.1 of the Territ Let (that is, 72 hours free of use of the operating area, changing for use of the operating area from day 4 ones The minimum volume of 25 thousand containers will be accounted during the period of validity of the offer; that is, between January 03, 2025 until December 31, 2025.

If, during weekly or monthly reviews of the volume of empty containers exported by shipping lines, it is found that the shipping line will not reach the minimum volume required during the annual period, then the offer to that shipping line will be cancelled.

10.2 Discount on the standard empty service tariff for lines that move a volume equal to or greater than 125,000 TEUs between full and empty.

Temporary offer. To container shipping lines that load (export) and unload (import) and unload (import) a volume equal to or greater than 125,000 TEU (between full and empty) during the months of May 2025 to August 2025 (quarterly period), a discount of USD 135 will be granted to empty 20-tox containers and a discount of USD 205 to empty 40-fox containers, with respect to the standard service fariff established in subsections 1.1.2.9 and 1.1.2.10 of the Tariff List. These discounts will be granted for containers moved between July 2025 and October 2025. These discounts will be granted up to a maximum of 1,600 empty 20-fox containers and 17,797 empty 40-fox containers.

- Effective: The offer may be requested from July 24, 2025 to July 30, 2025, and is valid until October 30, 2025.

- Lines that have loaded/unloaded a minimum of 80,000 TEUs in the last 2.5 months (prior to the start of the offer) will be eligible for this offer.
- Discounts apply to containers moved during the months of July 2025 to October 2025, but there is a maximum number of containers to which discounts will be granted.

- This offer will be subject to space availability confirmed by APMTC

### 10.3 Discount on the standard empty container service rate applicable to lines that exceed the container volume established in their service proformas by more than 3,000 containers.

- Target Group: Users (shipping lines or their representatives) who exceed the volume established in their container service pro-forma by the Multipurpose North Terminal by more than 3,000 contain

- Temporary offer: Shipping lines that exceed their committed volume of full and empty containers by more than 3,000 containers during the period from January 2025 to August 2025 will be granted a discount on the standard 40-foot empty container service tariff established in subsection 1.1.2.10 of the Tariff List. This discount amounts to USD 212 per empty 40-foot export container. This discount applies to empty been moved during the period from January 2025 to August 2025.

- Effective: The offer may be requested from July 24, 2025 to July 30, 2025, and is valid until August 30, 2025.

### - Requeriments:

- This offer is not available to lines that have applied to offer 11.2. This offer will be available to lines that, on the date the promotion is requested, have exceeded the volume established in their pro-forms by more than 3,000 conte
- The USD 212 discount applies to up to a maximum of 4,692 empty 40-foot containers.
- Discounts apply to containers moved during the aforementioned period; however, they do not apply to empty export containers that have received or accessed other discount offers for empty containers during 2025.

- This offer will be subject to snace availability confirmed by APMTC

### (Group 11) SPECIAL OFFER FOR BREAK BULK CARGO - TEMPORARY DEPOSIT OF APM TERMINALS CALLAO

### Dear customers and users

APM Terminals Callao (APMTC) has set a series of special offers for the special service "Integrated Temporary Depot Service" for treak bulk cargo. It should be noted that each of the special offers is independent (it is not cumulative); for example, if the user chooses promotion 11.4 he would not be able to apply to promotion 11.1, 11.2 or 11.3. Special offers are governed by the following the special offers is independent (it is not cumulative); for example, if the user chooses promotion 11.4 he would not be able to apply to promotion 11.1, 11.2 or 11.3. Special offers are governed by the following the special offers are governed by the

### 11.1 Extension of free days applicable to break bulk cargo in general and to subsection 2.4.1.1 of the Tariff List

erators, etc.) of break bulk cargo in general and that demand the integrated service of temporary storage with 10 days of use of the operational area (subs

- Segment: Users entering large volumes break bulk cargo at APM Terminals Callao Temporary Depot (DT 3014)

- Temporary offer: Users who guarantee minimum amounts of more than 2,500 and 3,500 bonnes using the APMTC's Temporary Depot will be granted an extension of the operating area use (taking into account the 10 days including in the Integrated Service of Temporary Depot sections 2.4.1.1 of the Tarff List), according to the table below

(i) ≥ 2,500 to 3,499 tonnes = 5 additional days off (on top of the 10 already granted in subsection 2.4.1.1 of the Tariff List)

(ii) ≥ 3,500 tonnes = 10 additional days off (on top of the 10 already granted in subsection 2.4.1.1 of the Tariff List)

- Effective: Users may apply or request this promotion from August 02, 2023 until December 31, 2024, and from January 3, 2025, to March 31, 2025, and from April 22, 2025, to June 30, 2025.

-Requesiments: The minimum volume of 2,500 or 3,500 tonnes will be accounted for annually; that is within heelve (12) months from the date of the agreement (between APMTC and the user) to commence the accounting of volumes (for example and as a hypothetical case, if the user's promotion request is accepted and it is agreed that the accounting will commence on September 1, 2023, then the accounting will be carried out until August 31, 2024, if it is fair the purpose, users should submit to APMTC's commencial sears the projections of three offers have not been reached, APMTC will change the list price set out in absolated 2-12 of the Teal Teal (see of a tes to 14.25 of the Teal Teal Center of the Period and the contract of the price set of the heaptical formation from the teal the minimum volumes required for the application of these offers have not been reached, APMTC will change the list price set out in absolated 2-12 of the Teal Teal Center of the Period offers and the price of the Indignated Teal Center of the Period Teal Center of the Per

### 11.2 Discount in price and extension of free days applicable to non-hazardous break bulk cargo in general and to subsection 2.4.1.1 of the Tariff List (integrated temporary depot service that includes 10 days of use of operating area)

ses amd Exporters/forwarders or its representatives – customs brokers, logistics operators, etc.) of break bulk cargo in general not dangerous that demand the Temporary Depot Integrated Service with 10 days of use of operative area (subsection 2.4.1.1 of the Tariff List) - Target group: Users (importers/co

Segment: Users entering large volumes break bulk cargo at APM Terminals Callao Temporary Depot (DT 3014)

- Emptorary offer: Users who guarantee minimum amounts of 5 thousands, 7 thousands and 10 thousands from such granted and such granted and second in price established in subsection 2.4.1.1 of the Tariff List and an extension of use of free operating area that will be in addition to the 10 days included in the Integrated Temporary depot Service (subsection 2.4.1.1 of the Tariff List), according to the label below.

(i) ≥ 5,000 to 6,999 tonnes = Discount of USD 2 per ton on the price of subsection 2.4.1.1 of the Tariff List and 10 additional free days (to the 10 already granted in subsection 2.4.1.1 of the Tariff List)

(ii) ≥ 7,000 to 9,999 tonnes = Discount of USD 4 per ton on the price of subsection 2.4.1.1 of the Tariff List and 10 additional free days (to the 10 already granted in subsection 2.4.1.1 of the Tariff List)

(ii) ≥ 10,000 tonnes = Discount of USD 6 per ton on the price of subsection 2.4.1.1 of the Tariff List) and 10 additional free days (to the 10 already granted in subsection 2.4.1.1 of the Tariff List)

- Effective: Users may apply or request this promotion from August 02, 2023 until December 31, 2024, and from January 3, 2025, to March 31, 2025, and from April 22, 2025, to June 30, 2025.

The minimum volumes of 5 thousand, 7 thousand, 7 thousand or 10 thousand from serving the accounted for over a period of nine (09) months; that is, in the 09 months following the date of the agreement (between APMITC and the user) to start the accounting of the volumes (for example and as a hypothetical case; if the user's promotion request is accepted and it is agreed to start accounting on September 1, 2023, then the accounting will be made until May 31, 2024). For this purpose, users must submit to the commercial area of APMITC the projections of lons to be mobilized in the nine-month period indicated above, in order to determine the availability of space in the terminal.

in the event that the user fails to comply with entering the committed volume. Or break bulk cargo into the DT 3014 within 09 months, an additional period of three (03) months may be granted to comply with the committed volume. This additional period will be granted as long as it has been verified that, in the first 09-month accounting period, the user has entered into the DT (3014) 100% of the bear three (03) months may be granted to comply with the committed volume. This additional period will be granted as long as it has been verified that, in the first 09-month accounting period, the user has entered into the DT (3014) 100% of the bear three (03) months may be granted to comply with the committed volume. This additional period will be granted to comply with the committed volume.

If, after the end of the nine (09) months of accounting (or the 12 months in the cases that apply the extension of term), it is verified that the minimum volumes required for this offer has not been read the its price established in subsection 2.4.1.2 of the Tariff List (Use of area for days 11 – 20 of the integrated temporary depot service) in force at the moment of the verification.

In case, the price of subsection 2.4.1.1 of the Tariff List is modified, the discounts of USD 2, USD 4 or USD 6 per ton apply to the modified price and in force at the time of invoicing See invoicing details in the in the Regulation of Tariffs and Commercial Policy of APM Ter

- This offer will be subject to space availability confirmed by APMTC

### 11.3 Discount in price and extension of free days applicable to high density break bulk cargo that can be stacked and to subsection 2.4.1.1 of the Tariff List (integrated temporary depot service that includes 10 days of use of operating area)

- Target group: Users (importers/consignees amd Exporters/forwarders or its repre ntatives – customs brokers, logistics operators, etc.) of high-density and stackable break bulk cargo (such as wood, ceramics, tiles, and others) that de mand the Temporary Depot Integrated Service with 10 days of use of operative area (subse

- Temporary offer: Users who guarantee minimum amounts of equal or more than 10 thousand bonnes using the APMITC's Temporary Depot, will be granted a discount of USD 9 per bon on the price established in subsection 2.4.1.1 of the Tariff List, an extension of 10 additional free days of use of the operating area with respect to the free days granted in subsection 2.4.1.1 of the Tariff List, an extension of 10 additional free days of use of the operating area with respect to the free days granted in subsection 2.4.1.1 of the Tariff List, an extension 2.4.1 of the Tariff List, an extension of 10 additional free days of use of the operating area with respect to the free days granted in subsection 2.4.1.1 of the Tariff List, an extension of 10 additional free days of use of the operating area with respect to the free days granted in subsection 2.4.1.1 of the Tariff List, an extension of 10 additional free days of use of the operating area with respect to the free days granted in subsection 2.4.1.1 of the Tariff List, an extension of 10 additional free days of use of the operating area with respect to the free days granted in subsection 2.4.1.1 of the Tariff List, an extension of 10 additional free days of use of the operating area with respect to the free days granted in subsection 2.4.1.1 of the Tariff List, an extension of 10 additional free days of use of the operating area with respect to the free days granted in subsection 2.4.1.1 of the Tariff List, an extension of 10 additional free days of use of the operating area with respect to the free days granted in subsection 2.4.1.1 of the Tariff List, an extension of 10 additional free days of use of the operating area with respect to the free days granted in subsection 2.4.1.1 of the Tariff List, an extension of 10 additional free days of use of the operating area with respect to the free days of use of the operating area with respect to the operating area with r

- Effective: Users may apply or request this promotion from October 06, 2022 until December 31, 2024, and from January 3, 2025, to March 31, 2025, and from April 22, 2025, to June 30, 2025.

This offer is only applicable to goods that can be stacked on at least three levels. If the merchandise can only be stacked on two levels, the offer does not apply. APMTC will define which goods are stackable up to three levels.

The minimum volume of 10 thousand tonnes will be accounted for in an annual period; that is, within 12 months of the date of the agreement, between APMTC and the user/ to start accounting for the minimum volume (for example and as a hypothetical case; if the user's promotion request is accepted and it is agreed that accounting will start on November 1, 2022, then accounting will be made until 30 October 2023). For this purpose, users should submit to the APMTC commercial area the projections of tonnes to be mobilized in the yearly period indicated above, in order to determine the availability of space in the terminal.

If, after the end of the 12 months of accounting, it is verified that the minimum volumes required for this offer has not been reached, APMTC will charge the fixt price established in subsection 2.4.1.1 of the Tariff List (integrated service of temporary storage of break bulk cargo that includes 10 days of use of operative area) as well as the prices for the use of the operating area from day 11 to 30 that has been used (prices of subsections 2.4.1.2 and 2.4.1.3 of the Tariff List) in force at the moment of the verification.

In case, the price of subsections 2.4.1.1 and 2.4.1.3 of the Tariff List is modified, the discounts of USD 9 per ton or USD 2.70 per ton iday apply to the modified price and in force at the time of invoicing. See invoicing details in the in the Regulation of Tariffs and Commercial Policy of APM Terminals Callao

- This offer will be subject to space availability confirmed by APMTC

# 11.4 Price discount applicable to break bulk cargo not dangerous and to subsection 2.4.1.1 of the Tariff List (integrated temporary depot service that includes 10 days of use of operating area)

- Target group: Users (importers/consignees amd Exporters/forwarders or its representatives - customs brokers, logistics operators, etc.) of general break bulk cargo not dangerous that demand the Temporary Depot Integrated Service with 10 days of use of operative area (subsection 2.4.1.1 of the Tariff List)

- Segment: Users entering large volumes break bulk cargo at APM Terminals Callao Temporary Depot (DT 3014)

- Temporary offer. Users who guarantee minimum amounts of more than 1.5 thousands, 2.35 thousands, 2.35 thousands, 3.3 thousands and 4 thousands tonnes using the APMTC's Temporary Depot, will be granted a discount in price established in subsection 2.4.1.1 of the Tariff List and an extension of use of free operating area that will be in addition to the 10 days included in the Integrated Temporary depot Service (subsection 2. 4. 1. 1. of the Tariff List), according to the table below:

(i) ≥ 1,500 to 2,349 tonnes = Discount of USD 2 per ton on the price of subsection 2.4.1.1 of the Tariff List and 10 additional free days (to the 10 already granted in subsection 2.4.1.1 of the Tariff List)

(ii) ≥ 2.350 to 3.299 tonnes = Discount of USD 4 per ton on the price of subsection 2.4.1.1 of the Tariff List and 10 additional free days (to the 10 already granted in subsection 2.4.1.1 of the Tariff List)

(ii) ≥ 3,300 to 3,999 tonnes = Discount of USD 6 per ton on the price of subsection 2.4.1.1 of the Tariff List and 10 additional free days (to the 10 already granted in subsection 2.4.1.1 of the Tariff List) = Discount of USD 6 per ton on the price of subsection 2.4.1.1 of the Tariff List and 20 additional free days (to the 10 already granted in subsection 2.4.1.1 of the Tariff List)

(iv) ≥ 4,000 tonnes - Effective: The users may apply or request this offer from January 3, 2025, to March 31, 2025, and from April 22, 2025, to June 30, 2025.

The volumes of the offers will be counted during the period of validity of the offer; that is, it will be counted between January 3, 2025, to March 31, 2025. For this offer there is no prior agreement or commitment of movements by the user against APM Terminals Callao

In case, the price of subsection 2.4.1.1 of the Tariff List is modified, the discounts of USD 2, USD 4 or USD 6 per ton apply to the modified price and in force at the time of invoicing. See invoicing details in the in the Regulation of Tariffs and Commercial Policy of APM Terminals Callso.

For users who have exceeded 10,000 tons during the offer's accounting period, they will be granted 5 additional days of free operational area use, in addition to the 20 or 30 days granted with the offers. These 5 additional days will apply to the tons that have entered the storage area since October 1, 2024, and remain in that area until December 15, 2024

- This offer will be subject to space availability confirmed by APMTC

# (Group 12) OFFER IN EXTENSION OF FREE DAYS FOR JOINT MOVEMENT OF BREAK BULK CARGO AND DRY CONTAINERS IN APM TERMINALS CALLAO

# Door crietomore and risore

APM Terminals Calao (APMTC) has set an offer, which involves an extension of days free of use of the operating area, for those users who move break bulk cargo and dry containers by the DT 3014. This offer is governed by the following term

- Target group: Users entering both break bulk cargo and dry containers into APM Terminals Callao Temporary Depot (DT 3014)

- Temporary offer. Any user who annually moves through DT 3014, minimum volumes of 3 thousand tons of breakbulk cargo, and 1,600 mportle-port containers, will be granted 20 days of free use of the operating area for the breakbulk cargo, which will be in addition to those 10 days of use of the operating area provided by the integrated 2.4.1.1 of the Tariff List); that is, the user will have 30 days of use of the operating area for the break bulk cargo, which will be in addition to those 10 days of use of the operating area for the break bulk cargo. The care of operating area for the break bulk cargo, which will be in addition to those 10 days of use of the operating area for the break bulk cargo. The care of operating area for the break bulk cargo, which will be in addition to those 10 days of use of the operating area for the break bulk cargo.

- Effective: The users may apply or request this offer from August 02, 2023 until December 31, 2024, and from January 3, 2025, to March 31, 2025, and from April 22, 2025, to June 30, 2025.

The minimum volumes will be counted annually, that is to say, in the 12 months following the date of the agreement (between APMIC and the user) to start counting the volumes (for example and as a hypothetical case; if the user's request for promotion is accepted and it is agreed that the posting on December 1, 2023, then the accounting will be made until Nove this purpose, users must submit to the commercial area of APMIC the projections of break bulk cargo and dry container that will be imported exported in the annual period indicated above, in order to determine the availability of space in the terminal.

APMTC will carry out a review every two (IZI) combine the volume of containers and break-bulk cargo modified by the user requested in the summal.

APMTC will carry out a review every two (IZI) combine the volume of containers and break-bulk cargo modified by the user requested from the user. If the sort modified by the user migrated in the user will not be able to meet the required annual volume, then APMTC may cancel the offer. If it is writer that the user has not been annual volume (here a bulk cargo and containers) required by the offer, the offer will be cancelled from the user, if the first review verifies that the user modified less than 10% of the annual volume (here a bulk cargo and containers) required by the offer, the offer will be cancelled from the user, if the total veriew verifies that the user modified less than 20% of the annual volume (here a bulk cargo and containers) required by the offer, the offer will be cancelled from the user, if in the furth review it is verified that the user modifies less than 20% of the annual volume (hereal bulk and containers) required by the offer, the offer will be cancelled from the user, if in the furth review it is verified that the user modifies less than 20% of the annual volume (hereal bulk and containers) required by the offer, the offer will be cancelled from the user, if in the furth review it is verified that the user modifies less than 20% of the annual volume (hereal bulk and containers) required by the offer, the offer will be cancelled from the user, if in the furth review it is verified that the user modifies less than 20% of the annual volume (hereal bulk and containers) required by the offer, the offer will be cancelled from the user, if in the furth review it is verified that the user modifies less than 20% of the annual volume (hereal bulk and containers) required by the offer, the offer will be cancelled from the user, if it is the user modifies less than 20% of the annual volume (hereal bulk and containers) required by the offer, the offer

- This offer will be subject to space availability confirmed by APMTC

# (Group 13) OFFERS FOR RO-RO CARGO MOBILIZED BY APM TERMINALS CALLAO

# Dear customers and users

# 13.1 Discount or price reduction for light vehicles mobilized through APM Terminals Callao (as temporary depot or port terminal)

- Target group: Users who enter light vehicles to the Temporary Depot of APM Terminals Callac (DT 3014) or who demand use of operational area as a port terminal up to day 10. Temporary offer: Any user who mobilizes light vehicles through APM Terminals Callao, either as DT 3014 or as Port Terminal, will be charged the following offers prices:

(i) Through APMTC's Temporary Depot (DT 3014), a offer price of USD 75 per ton will be charged for the special integrated temporary depot service for ro-ro cargo (s. (ii) As a Port Terminal, a offer price of USD 32.20 per ton will be applied for the use of operational area until day 10 (subsection 3.3.1.2 of the Tariff List). ion 3.4.1.1 of the Tariff List).

- Effective: Users may apply or request the promotion 1) from October 16, 2022 until December 31, 2024, from January 3, 2025, to March 31, 2025, and from April 11, 2025, b. June 30, 2025. While promotion ii) applies between November 8, 2022 and December 31, 2024, from January 3, 2025, to March 31, 2025, and from April 11, 2025, b. June 30, 2025.

- Requeriments: With light vehicles we refer to the definition established in Supreme Decree No. 25-2008-MTC, but excluding trailers, chassis and vehicles that exceed 2 tons per unit. - This offer will be subject to space availability confirmed by APMTC

### 13.2 Offer for users who address machinery (ro-ro cargo) to APMTC's Temporary Depot (DT 3014) - Target group: Users who enter machinery to APMTC's Temporary Depot (DT 3014)

- Temporary offer: Any user who address machinery (in the form of ro-rocargo) to DT 3014, will be charged the following offer:

(i) Between 500 and 1,999 bornes = 20 additional days of free use of the operational area with respect to the one granted in subsection 3.4.1.1 of the Tariff List (which is 10 free days); that is, a total of 30 days of free operating area use will be granted.

2 St bossands forms = USD 5 discount on the current price of subsection 3.4.1.1 of the Tariff List and 20 additional days of free use of the operational area with respect to the one granted in subsection 3.4.1.1 of the Tariff List (which is 10 free days); that is, a total of 30 days of free operating area use will be granted.

- Effective: The users may apply or request this offer from February 24, 2023 until March 31, 2024, from May 2, 2024 to December 31, 2024, from January 3, 2025, to March 31, 2025, and from April 22, 2025, to June 30, 2025.

Machinery refers to those units that are heavy and are not decicated to the transport of persons or goods, as defined in national legislation (such as Supreme Decree No. 25-2008-AITC and others); that is, light or heavy vehicles are not eligible for the offer.

The minimum volumes will be counted annually, that is to asy, in the 12 months following the date of the agreement (between PARIVO. and the user) to start counting the volumes (for example and as a hypothetical case; if the user's request for promotion is accepted and it is agreed that the posting on March 1, 2023, then the accounting will be made until February 28, 2024). For this purpose, uses must submit to the commercial areas of ARIVO the projections or the memory and reveal of the r

### 13.3 Discount or price reduction for light vehicles mobilized through APM Terminals Callao (as port terminal) and are collected in storks (car transporter)

- Target group: Users moving light vehicles through the Multipurpose North Terminal

- Temporary offer. Any user who mobilizes light vehicles, requests the standard service to ro-ro cargo (cargo portion) and the use of an operating area (such as Port Terminal) and withdraws light vehicles from the Multipurpose North Terminal by means of specialized trucks for the transport of these vehicles (called storks), will be charged the following promotional price:

Adiscount of USD 70 per tonne will be applied for standard service (cargo portion) to re-or cargo (subsection 3.1.2.1 of the Tariff List) and use of operational area use set out in subsection 3.3.1.2 of the Tariff List (which, on the date of publication of this promotion, means charging the offer price of USD 192.80 per ton), will not apply, but the combined price of these two services will be subject to the discount of USD 192.80 per ton).

- Effective: The users may apply or request this offer from March 09, 2023 until December 31, 2024, and from January 3, 2025, to March 31, 2025, and from April 11, 2025, to June 30, 2025.

- Requeriments

In the event of a change in the standard service tariff or the operating area usage price, the amount of the discount does not change, but the amount that will be charged for the offer will change.

The user who requests the promotion may use the operating area up to day 5 at the latest, the user decides if he stays less than the maximum allowed, but if he stays less than 5 days and has requested the promotion, the offer price will be applied to him. Users have up to 48 hours before the vessel's ETB to decide whether to request only the standard service or apply for the offer.

With light vehicles we refer to the definition established in Supreme Decree No. 25-2008-MTC, but excluding trailers, chassis and vehicles that exceed 2 tons per unit.

In case users pick up their vehicle, using storks (car transporter), on day 6 or later, the price set out in subsection 3.3.1.2 of the Tariff List (which at the date of publication of this promotion is USD 90 per ton) will apply.

- This offer will be subject to space availability confirmed by APMTC

# (Group 14) DISCOUNT OR ADDITIONAL REDUCTION IN TARIFF FOR LIQUID BULK CARGO - VEGETABLE OILS

APM Terminals Callao (APMTC) has arranged that, for vegetable oils that are unloaded in the form of liquid bulk cargo, a discount will be applied for the standard liquid bulk service (subsection 5.1.2.1 of the Tariff Schedule) if meet certain conditions. This discount is governed by the following terms:

(i) Discount of USD 2.3 per ton on the maximum liquid bulk standard service tariff (in force at the time of invoicing) for vegetable oil discharged through pipelines (no use of lecterns) This offer will be valid from July 01, 2025 to December 31, 2025.

# (Group 15) DISCOUNT OR ADDITIONAL REDUCTION IN TARIFF FOR SOLID BULK CARGO - CLEAN GRAINS

APM Terminals Cale (pulmC) has established that users who unload solid bulk wheat in a volume exceeding 180,000 tons during the months of May 2025 to August 2025 (four-month period) will be charged a discount of USO 2 per ton for the standard solid bulk service (subsection 4.1.2.1 of the Tariff List). This latest discount will be applied up to a maximum volume of 90,000 tons. This discount will be valid and will be applied from July 24, 2025, to November 30, 2025.

# (Group 16) DISCOUNT FOR IMPORT CONTAINERS FOR EARLY PICKUP

APM Terminals Calao (APMTC) has arranged that users who pick up their import containers on the 1st or 2nd day after entering the container yard will receive discounts on the standard service tariff for the cargo-portion. Details of this temporary offer are as follows:

- Target group: Users who import import containers through the Terminal.

- Temporary offer: Any user who removes their import containers on day 1 or day 2 after their arrival at the container yard will receive a discount on the standard service tariff for the cargo-portion (as indicated in subsections 1.1.2.1, 1.1.2.2, 1.1.2.5, and 1.1.2.6 of the Tariff List). The discount amount will be USD 50 if the container is pick up on day 1 and USD 25 if it is pick up on day 2.

- Effective: This offer will be valid from September 23, 2025 to October 05, 2025.

Day 1 will be counted from the date and time of entry into the container yard until 24 hours on the same calendar day. Day 2 will be counted from 00 hours on the following calendar day until 24 hours on the following calendar day.

It should be mentioned that these accounting days for the present offer fifter from the way storage stay days are counted for the purposes of the free days offered by the standard service (and as well as what would have to be paid if they exceed the free days granted by the standard service).

Published and updated on 23/09/2025 Applicable as from 23/09/2025 and 30/09/2025

Special tariffs (maximum tariffs reductions) for regulated services, unless otherwise expressly indicated in the Tariff List, will be in force from July 01, 2025 to December 31, 2025.

For a greater detail of the scope and description of our services, commercial policies and surcharges, see the Regulation of Tariffs and Commercial Policy of APM Terminals Callao S.A.: https://www.apmterminals.com/en/callao/customer-zone/tariffs